

Oldenburg Discussion Papers in Economics

The Relationship between Age and Subjective Well-Being:

Estimating Within and Between Effects Simultaneously

Philipp Biermann

Jürgen Bitzer

Erkan Gören

V – 421-19

March 2019

Department of Economics University of Oldenburg, D-26111 Oldenburg

The Relationship between Age and Subjective Well-Being: Estimating Within and Between Effects Simultaneously *

Philipp Biermann[†] Jürgen Bitzer[‡] Erkan Gören[§]

Otto von Guericke University Magdeburg Carl von Ossietzky University Oldenburg Carl von Ossietzky University Oldenburg

This Version: March 2019

Abstract

Based on data from the German Socio-Economic Panel (SOEP), we used a correlated random effects econometric framework to simultaneously estimate the within and between effects of age on subjective well-being. The proposed approach overcomes the ambiguity in the relationship between age and subjective well-being reported in a series of studies based on cross-sectional and/or longitudinal panel data. Our results suggest that a cubic-type functional relationship between well-being and age fits the data best, leading to highly significant coefficient estimates associated with the age variables, and consistent within and between effects of age on subjective well-being. A linear or quadratic functional relationship between well-being and age is not empirically supported, as the between and within estimates of age on well-being differ significantly from each other. The main findings are robust to the inclusion of a broad range of individual-level sociological, demographic, and economic controls, and to the inclusion of various interviewer controls such as survey experience, survey type, and interviewer fixed effects.

Keywords: Subjective Well-Being, Life Cycle Happiness, Cohort Effects, Mundlak Approach, Correlated Random Effects, Fixed Effects, Between- and Within-Person Effects

JEL Classification Numbers: I31, C21, C23, J10

^{*}We would like to thank Ingo Geishecker, Heinz Welsch, the session participants at the Annual Meeting of the Verein für Socialpolitik (VfS) 2018, University of Freiburg, and the seminar participants at the Carl von Ossietzky University Oldenburg 2018 for helpful comments and suggestions. All remaining errors are our own.

[†]Otto von Guericke University Magdeburg, Faculty of Human Sciences, Institute of Sociology, Building 40, 39104 Magdeburg, Germany, Tel.: +49-391-67-56612, e-mail: philipp.biermann@ovgu.de.

[‡]Carl von Ossietzky University Oldenburg, School of Computing Science, Business Administration, Economics, and Law (Faculty II), Institute of Economics, Building A5, 26111 Oldenburg, Germany, Tel.: +49-441-798-4217, e-mail: juergen.bitzer@uni-oldenburg.de.

[§]Carl von Ossietzky University Oldenburg, School of Computing Science, Business Administration, Economics, and Law (Faculty II), Institute of Economics, Building A5, 26111 Oldenburg, Germany, Tel.: +49-441-798-4292, e-mail: erkan.goeren@uni-oldenburg.de.

1 Introduction

Over the last few decades, the relationship between age and individual well-being¹ has been the subject of intense discussion in various research fields, including economics, gerontology, psychology, and sociology. Researchers from the different fields have provided a range of theoretical explanations for this relationship and tested them empirically. The functional relationships found between well-being and age in empirical studies are manifold. The findings range from U-shaped and inverted U-shaped relationships to linear or cubic relationships or no relationship at all. Moreover, theoretical explanations can be found for virtually any empirical relationship. The manifold number of theoretical explanations arise from the multidimensionality of channels by which age might affect subjective well-being (SWB). Those theoretical models range from consumption-smoothing (Fisher, 1930; Modigliani and Brumberg, 1954) and hyperbolic discounting (Frederick et al., 2002) models in economics to models of set point theory (Diener and Lucas, 1999; Lucas, 2007) and the hedonic treadmill (Kahneman, 1999) in psychology, to the satisfaction paradox (Mroczek and Kolarz, 1998) and socio-emotional selectivity theory (Munsey, 2007; Charles and Carstensen, 2009) in gerontology.²

A striking characteristic of the empirical literature is the persistence of arbitrary results concerning the functional form found. Most studies using cross-sectional data and analyzing the between-person variation of age find a U-shaped relation with a low around midlife (Blanchflower and Oswald, 2004, 2008; Graham and Ruiz Pozuelo, 2017; Hayo and Seifert, 2003; Lang et al., 2011). Studies by Mroczek and Spiro (2005) and Easterlin (2006) report an inverted U-shape based on very special datasets such as the US Department of Veterans Affairs (VA) Normative Aging Study (NAS) and the synthetic panel of the US General Social Survey (GSS). The studies by Frijters and Beatton (2012) and Kassenboehmer and Haisken-DeNew (2012) show that the U-shaped effect is persistent in a cross-sectional framework, while it disappears when the empirical analysis includes panel fixed effects to control for unobserved individual-level heterogeneity. Meanwhile, Kassenboehmer and Haisken-DeNew (2012) found that age has no effect on subjective well-being if the regression model accounts for the number of years the respondent has been part of the survey. Finally, in an empirical analysis of British and German representative household surveys, Wunder et al. (2013) report that subjective well-being follows a wave-like pattern over the life course: life satisfaction declines until midlife, then starts to increase, and eventually seems to decline again among the oldest old. The methodology used is quite different from previous approaches on the topic, as it implements a semi-parametric regression model with penalized splines (P-splines) to model the non-linear pattern of well-being across the life span. Overall, while theoretical explanations of the lifetime pattern of well-being vary widely across the scientific disciplines, the empirical evidence remains mixed even within the disciplines and especially in the literature on economics.³

In our review of the relevant literature, we identified three key methodological issues that we used to differentiate the empirical studies on lifetime patterns of well-being into groups. The first and most important methodological issue is the use of cross-sectional and/or longitudinal panel data. While in the first case, variation between survey respondents

¹The terms well-being, happiness, and life satisfaction are used interchangeably in this article.

 $^{^{2}}$ Both, consumption smoothing and hyperbolic discounting rely on the assumption that expected utility equals experienced utility (Kahneman et al., 1997), which according to the theoretical framework in Ramsey (1928), yields increasing/constant/decreasing utility profiles depending on the difference between the time preference rate and the interest rate.

 $^{^{3}}$ See López Ulloa et al. (2013) for a review of the main findings in the relevant theoretical and empirical literature on the relationship between well-being and age.

is used to identify the effect of age on well-being over the life span, in the second case, it is the time variation within survey respondents. In contrast to cross-sectional and/or repeated cross-sectional data, the use of longitudinal panel data enables the inclusion of panel fixed effects in the regression model. The latter methodological approach provides coefficient estimates that are robust to individual-level heterogeneity due to the omission of key time-constant unobserved factors (for example, specific personality and/or genetic traits that might be correlated with both reported life satisfaction and longevity).

The second crucial difference between the empirical studies pertains to the specification of the age variable. Studies either employ indicator variables for each individual age category or pool individual observations into different age groups (defined for 10-, 5-, or 3-year intervals). Despite the functional flexibility of such an approach, it introduces the well-known identification problem regarding the simultaneous inclusion of birth cohort, time, and age effects into the regression model (Blanchflower and Oswald, 2008; Baetschmann, 2014). The majority of studies use higher-order polynomials of the age variable to examine the lifecycle pattern of well-being. For example, the U-shaped and/or inverted U-shaped relationship between well-being and age is typically tested using a quadratic specification of the age variable. Despite the functional simplicity of this approach, the use of higher-order polynomials predetermines the results to some extent. Specifically, the use of a quadratic function in age explicitly rules out more complicated relationships (e.g., a cubic association) between well-being and age.

The third important methodological difference between the studies is the selection of sociological, demographic, and economic individual-level controls that might confound the empirical relationship between well-being and age. Most researchers agree that factors such as household income, education, employment, health, marital status, gender, immigration status, number of children, birth cohort effects, panel attrition controls, survey-type fixed effects, and individual fixed effects (depending on the type of data employed, i.e., cross-sectional and/or longitudinal panel data) should be included as standard control variables in the regression model. However, there is little consensus among researchers as to whether the empirical analysis should explicitly account for time effects, the respondent's survey experience, and the possible confounding influence of the interviewer (e.g., interviewer gender and/or experience).

Overall, the plethora of associations between well-being and age (e.g., U-shape, flat, or wave-like pattern) could be a true empirical finding or could simply reflect a statistical artefact caused by study-level methodological differences in the type of data sets used, the econometric methods applied, the functional forms tested, and the set of control variables included. Given this heterogeneity, it should be clear that the literature is far from being able to draw final conclusions about the evolution of well-being over the life span.

Based on individual-level panel data from the German Socio-Economic Panel (SOEP) during the period 1992 to 2016, we propose a unified empirical framework to examine the impact of the aforementioned methodological issues on the relationship between well-being and age. To address the first methodological issue, we used a hybrid version of the correlated random effects (CRE) framework, first proposed by Mundlak (1978), to examine the within and between effect of age on individual well-being simultaneously. This estimation framework has proven particularly useful to relax the zero correlation assumption between variables that vary across time and the individual-level unobserved heterogeneity in a standard random effects (RE) estimation approach. Extending the regression model through the inclusion of individual-level group means of the time-variant explanatory variables and estimating this augmented model under the RE assumption effectively controls for individual-level heterogeneity. In contrast to the standard fixed effects regression model, the CRE framework is capable of disentangling the within and between effects of time-variant controls. It is worth mentioning that this estimation framework even allows the inclusion of individual-level controls that do not vary over time (e.g., gender) and that would otherwise be lost when using the fixed effects estimator. We addressed the second methodological issue regarding the proper specification of the age variable by using higher-order polynomials for age. This allowed us to test competing theoretical explanations about the lifecycle pattern of well-being (i.e., linear, flat, U-shaped, inverted U-shaped, or wave-like patterns). We examined the last methodological issue by including a large battery of individual-level controls that have been identified as main determinants of differences in individual life satisfaction both across time and across individuals. Beyond the aforementioned standard sociological, demographic, and economic factors, we also examined the robustness in the association between well-being and age to survey-specific conditions such as the respondent's experience answering survey questions, the type of survey conducted (e.g., by phone or personal interview), whether the interview was conducted by a male, and whether the interviewer had extensive experience surveying respondents.

The results suggest that a cubic functional specification in the age variable leads to consistent within and between effects of age on well-being over the entire life span. This finding is consistent with a wave-like pattern in subjective well-being over the life span. Furthermore, this specification is highly robust to the inclusion of a broad range of individual-level controls. This is in stark contrast to the regression results either using a linear and/or quadratic specification of the age variable, in which the corresponding within and between effects of age on well-being are contradictory and extremely sensitive to the inclusion of additional individual-level controls. It is worth mentioning that for the quadratic model specification, the U-shaped pattern in well-being is only present for the between-person dimension, whereas the within-person results would predict that the association is *inverted* U-shaped. Since this finding is based on the same number of observations and/or individuals observed at different points in time, we conclude that the U-shaped pattern in well-being reported in previous studies could be a statistical artefact caused by the cross-sectional dimension of the underlying survey data.

The remainder of this paper is organized as follows. Section 2 provides a review of the relevant literature on the empirical association between well-being and age, followed by a discussion of the key methodological differences across studies and their implications for the revealed association between well-being and age. Section 3 discusses the validity of the proposed estimation strategy. Section 4 describes the data and variables employed in the empirical analysis. Section 5 describes the main empirical results. Finally, Section 6 concludes by summarizing the main findings.

2 Relevant Literature

In this section, we provide a systematic literature review on the empirical relationship between individual life satisfaction and age. We identify substantial methodological differences across empirical studies (e.g., type of data, econometric estimators, age coverage of survey respondents, and a set of included individual-level controls, among other factors) that prevent any final conclusions about the true relationship between well-being and age. We conclude that the plethora of empirical findings on the lifecycle pattern of well-being (e.g., U-shaped, flat, or wave-like pattern) could be a true empirical finding or could simply reflect a statistical artefact due to study-level methodological differences.

2.1 The Functional Relationship Between Subjective Well-Being and Age

Although a large number of studies have examined the relationship between subjective well-being and age, the literature is far from being able to draw definitive conclusions on the topic. In order to gain a broad understanding of the main findings, Table 6 provides a systematic review of relevant studies on the well-being-age nexus. The table reveals that the majority of the studies report a U-shaped relationship between subjective well-being and age, with a low point occurring between the mid-30s and mid-50s. Life satisfaction seems to decline from young adulthood until midlife, when it reaches a turning point and then increases again. This result is consistent across various social survey databases from many different countries.⁴

Using data on 500,000 randomly sampled individuals, Blanchflower and Oswald (2008) found evidence of the U-shaped curve in well-being in separate regressions of 72 developed and developing nations in America, Europe, Latin America, and Asia. Similarly, Graham and Ruiz Pozuelo (2017) found that the U-shaped relationship between well-being and age holds for 44 out of 46 individual countries. The authors used nationally representative household surveys from the Gallup World Poll (GWP) conducted during the years 2005 to 2011.

Some scholars still have serious reservations, however, and argue that the U-shaped curve in well-being over the life span could be a pure statistical artefact caused by study-level methodological differences. A striking observation is that the U-shaped pattern is found mainly in studies using cross-sectional or repeated cross-sectional observations. To account for unobserved factors confounding the statistical relationship between well-being and age, most studies include a standard set of individual-level controls such as income, marital status, gender, and employment status in the regression equation. These studies are mainly interested in examining the "pure" age effect on subjective well-being, i.e., the effect after accounting for objective life circumstances that are associated with age, such as the number of children or declining health.

A small number of studies examine the "unadjusted" pattern of well-being over the life span (i.e., without controlling for any socioeconomic factors). For example, Easterlin (2006) from a synthetic panel analysis of the US General Social Survey (GSS) found an inverted U-shaped relationship between well-being and age. Their ordered logit regression analysis of individual well-being measures only accounts for a limited number of demographic controls such as cohort effects, gender, ethnic variables, as well as education status – factors that barely change over time.

Remarkably, the U-shaped curve in well-being becomes relatively flat in regression models that explicitly account for panel fixed effects in longitudinal studies. For example, Frijters and Beatton (2012) looked for a U-shaped in well-being using three national representative longitudinal household surveys, i.e., the German Socio-Economic Panel (SOEP), the British Household Panel Survey (BHPS), and Household, Income, and Labour Dynamics in Australia (HILDA). The authors report that all three national representative household surveys refute the U-shaped curve in well-being typically found in pooled OLS regressions once individual fixed effects are accounted for in the regression analysis. In a parallel study, Kassenboehmer and Haisken-DeNew (2012) also found no U-shaped curve in well-being in the SOEP data, additionally controlling for the possible confounding effect of interviewer characteristics.

 $^{^{4}}$ Another approach to analyze the relation between age and subjective well-being is to use information on mental distress indicators or the use of anti-depressants. Lang et al. (2011) found that mental distress follows an inverted U-shaped pattern throughout the life cycle and that it is more prominent in low-income than in high-income households in a pooled cross-section of countries. Blanchflower and Oswald (2016) showed that the use of anti-depressants reaches a maximum in midlife and declines afterwards.

There are two notable exceptions to the U-shaped pattern in well-being based on longitudinal data but using different methodological approaches. One of these, Van Landeghem (2012), was an analysis of the SOEP, and the other, Cheng et al. (2017), used longitudinal panels from four household surveys, the SOEP, BHPS, HILDA, and MABEL, and found consistent evidence of a U-shaped relationship between well-being and age. The methodology used in both studies is quite different from previous longitudinal analyses, as it rests on the assumption that the first derivative of a quadratic function (i.e., the U-shaped pattern in well-being over the life span) is linear with a positive slope. However, it is worth mentioning that neither study includes panel fixed effects estimates; instead, both used a simple graphical method to examine the functional relationship between the mean yearly *changes* in overall subjective well-being and age. A positive and statistically significant linear function appears to provide evidence that the unadjusted (raw) pattern in well-being is U-shaped over the life span. Following the narrative in Easterlin (2006), neither the study of Van Landeghem (2012) nor Cheng et al. (2017) include other control variables in the regression analysis. Thus, it remains unclear whether the main findings are subject to unobserved individual-level heterogeneity.

Several other studies have looked at a more complicated functional relationship between well-being and age. Wunder et al. (2013) applied a semi-parametric regression model with penalized splines (P-splines) to the British (BHPS) and German (SOEP) longitudinal data. The authors found that well-being over the life span has three stages. The first two stages are described accurately by the U-shaped pattern in well-being found in previous studies and seem to fit people's experience until their late 60s. Afterwards, in the third stage of life, a decline in well-being takes place among the oldest old in both households surveys, consistent with the hypothesis of declining life satisfaction due to a deterioration of health among older people. Such a wave-like pattern in well-being has been replicated in several other independent studies (Gwozdz and Sousa-Poza, 2010; Baetschmann, 2014; Wooden and Li, 2014), suggesting that the U-shaped functional relationship might be too restrictive to describe the entire pattern of well-being over the life span.

Overall, the above review suggests that the flat, (inverted) U-shaped, and wave-shaped lifecycle patterns in well-being reported in the literature could reflect true empirical findings or simply be the consequence of study-level methodological differences. In principle, each of the functional relationships between well-being and age can be justified on some theoretical grounds.

For example, the economic life-cycle theory predicts that life satisfaction would be flat over the entire life span because of inter-temporal optimization of agents at each point in time (Blanchflower and Oswald, 2008; Van Landeghem, 2012). However, no standard economic theory exists that might explain the non-linear relationship between well-being and age over the life span, and any economic modelling that generates a non-linear (e.g., U-shaped) association between the two outcomes would be rather ad-hoc.

There are two overlapping explanations that might be consistent with the U-shaped pattern in well-being over the life span. One explanation is that the decline in life satisfaction up to midlife and its increase thereafter (the phenomenon also known as "midlife crisis") might be caused by lower and higher adaption levels, respectively, to missed goals and opportunities throughout life (Frey and Stutzer, 2002; Frijters and Beatton, 2012).⁵ The second possibility is that the increase in life satisfaction after midlife could be explained by the fact that older people value the remaining years more

⁵This observation is consistent with neurological and psychological studies that argue that an individual's emotional reaction to missed chances and unmet aspirations decreases with age (Brassen et al., 2012).

because their friends are passing away (Blanchflower and Oswald, 2008).

Weiss et al. (2012) provided an alternative explanation of the U-shaped pattern in well-being. The authors showed that a similar U-shape exists among 508 great apes whose well-being was assessed by raters who had known the individual apes for at least two years. Their results imply that the U-shaped pattern in well-being is not uniquely human and that its origins are partly rooted in human biology beyond standard socioeconomic factors. These findings might help scientists across various disciplines to develop a theoretical framework that would explain the (possibly nonlinear) pattern in well-being over the entire life span.

The steep decline in well-being among the oldest old might be caused by a deterioration of health. Another explanation would be the *Effect of Impending Death*. Life satisfaction sharply declines in the years prior to death (Small and Beckmann, 1997). This effect is reported in studies that explicitly control for diseases that usually occur in around the same period, suggesting that the decline in life satisfaction among the oldest old is not entirely explained by a decrease in health status (Small et al., 2003). In combination with the U-shaped relationship between well-being and age, the latter argument would provide an explanation for why life satisfaction follows a wave-like pattern over the entire life span (Wunder et al., 2013).

In summary, given the heterogeneity of the studies in terms of (i) data design (e.g., cross-sectional, longitudinal panel, or repeated cross-sectional data), (ii) choice of econometric estimators (e.g., panel fixed effects, random effects, pooled OLS, ordered logit, or semi-parametric regression models), (iii) type of national household survey (e.g., country of origin, socio-demographic composition of survey respondents), (iv) definition of the age variable (e.g., continuous age function and/or step function for different age groups), (v) the choice of individual level controls (e.g., income, education, health, and marital status), (vi) the type of functional relationship tested (e.g., flat, U-shaped, or wave-shaped pattern), and (vii) the definition of the dependent variable (e.g., life satisfaction and/or general health questionnaire measure), a definitive conclusion about the pattern of well-being over the entire life span is difficult to draw. We contribute to the relevant literature by assessing the relative importance of the aforementioned methodological concerns that might confound the true relationship between well-being and age.

2.2 The Importance of Study-Level Methodological Differences

Beside the supposed functional relationship between well-being and age tested in the econometric estimations, the studies also exhibit broad heterogeneity regarding the choice of standard individual-level control variables. There is broad consensus in the empirical literature that sociological, demographic, and economic controls (e.g., income, household composition, employment, marital and health status) have to be included in the formal regression models to explain variation in well-being over time (Dolan et al., 2008). However, other studies refrain from controlling for social and economic factors in the regression models, since their focus is on the unadjusted (or raw) pattern of well-being over the life span (Easterlin, 2006; Easterlin and Sawangfa, 2007).

Studies can be further differentiated with respect to the inclusion of panel fixed effects that effectively account for unobserved individual-level heterogeneity across survey respondents. While the inclusion of panel fixed effects has become standard in studies using longitudinal data (Frijters and Beatton, 2012; Kassenboehmer and Haisken-DeNew, 2012), such an approach is technically not feasible when using cross-sectional or repeated cross-sectional data with a changing composition of survey respondents at each point in time (Stone et al., 2010; Graham and Ruiz Pozuelo, 2017). As the differences between the two methodological approaches are well known, we refrain from repeating the technical arguments at this stage. Taken together, however, the findings from cross-sectional and longitudinal panel data studies offer an important insight: The inclusion of panel fixed effects in the regression model clearly affects the empirical association between wellbeing and age. As discussed above, Frijters and Beatton (2012) have shown that the U-shaped pattern in well-being becomes rather flat if the regression model controls for the possible confounding effects of unobserved individual-level heterogeneity.

Another important difference among the various studies is the consideration of birth cohort and/or time effects. The argumentation is that even though well-being remains constant over time, survey respondents born in different years might report on average different levels of well-being. This observation might result from different socialization strategies during the time period in which survey respondents were raised (Van Landeghem, 2012). Thus, the U-shaped pattern in well-being could simply be a statistical artefact reflecting the different levels of reported life satisfaction of heterogeneous individuals raised in different cohorts. Indeed, birth cohort factors are significantly correlated with subjective well-being (Gwozdz and Sousa-Poza, 2010; Baetschmann, 2014; Clark and Oswald, 2006; Graham and Ruiz Pozuelo, 2017; Fukuda, 2013; Clark, 2007). However, the U-shaped pattern in well-being over the life span remains statistically significant. This implies that the U-shaped relationship between well-being and age cannot be entirely explained by the inclusion of individuals born in different years.

Recent studies have begun to investigate the influence of response behavior, i.e., determinants of which might bias the respondent's answers on individual subjective well-being. The existing studies tackle two dimensions of this issue: first, the respondent's experience, and second, characteristics of the interviewer. The first dimension reflects the possibility that respondents might become more aware or honest the more often they answer questions regarding their well-being (see Wooden and Li (2014) for panel conditioning). This aspect is usually captured by controlling for the amount of time an individual has spent in the panel (Frijters and Beatton, 2012; Baetschmann, 2014; Kassenboehmer and Haisken-DeNew, 2012). On the other hand, Wooden and Li (2014) argued that a long participation in a survey might cause growing disinterest and disengagement on the side of the respondent, leading to panel attrition. To control for this, they propose including a dummy that marks individuals who drop out in the next wave of the survey.

The second dimension considers the possibility that the respondent might answer differently depending on the interviewer who carries out the survey. Kassenboehmer and Haisken-DeNew (2012) therefore included the interviewer's gender and experience as control variables in their estimations. The results show that this leads to insignificant coefficients on age, which calls the impact of age on subjective well-being into question. Closely connected to this issue is the type of technique in which the interview was carried out, e.g., face-to-face or by phone, questionnaire filled out by interviewer or with the help of the interviewer, with or without an interpreter. Therefore, Wooden and Li (2014) controlled for the interview type carried out and found that they have an impact on the reported subjective well-being.

Overall, it has to be stated that although there is consensus regarding the inclusion of very basic sociological, demographic, and economic control variables, studies still differ in the specification of the key age variable, the inclusion of interviewer characteristics, and the use of appropriate econometric techniques (e.g., panel fixed effects versus pooled OLS regressions). The latter is partly explained by the availability of cross-sectional and/or longitudinal panel data, but some studies have argued against the inclusion of panel fixed effects in empirical analyses based on longitudinal panel data (Baetschmann, 2014; Cheng et al., 2017).

Study, Docion /Authors Date		Table 1: On the Relationship Between Well-Being and Age: A Review of the Relevant Empirical Literature Discrete Discrete Discrete Discrete Discrete	ng and Age: A Review of the Relevant En	npirical Literature	Chanzant al. (2017)
Study Design/Author, Date	Baetschmann (2014)	bianchnower and Oswald (2004)	Blanchhower and Oswald (2008)	Bianchnower and Oswald (2010)	Cheng et al. (2011)
Data Type	pooled cross-sectional data	pooled cross-sectional data	pooled cross-sectional data	pooled cross-sectional data	longitudinal panel data
Sample Size	304,856	38,000 people from the US and 57,000 peo- ple from the UK	observations range between 18,000 and 945,000, depending on the survey used	about 28,000	BHPS (98,320), HILDA (80,389), SOEP (185,169), MABEL (9,297)
Time Coverage	1984-2009	1972-1998	depending on survey used	2005, 2006, 2010	BHPS (1991-2008), HILDA (2001-2010), SOEP (1984-2008), MABEL (2009-2011)
Number of Individuals	38,197	N/A	N/A	N/A	N/A
Methodology	pooled OLS regressions	ordered logit regressions	ordered logit regressions	pooled OLS, probit, ordered logit regres- sion	pooled OLS regressions
Country	former West Germany	United States, Great Britain	various countries	27 European countries	Great Britain, Germany, Australia
Survey Type	German Socio-Economic Panel (SOEP)	US General Social Surveys (GSS) and UK Eurobarometer surveys	US General Social Surveys (GSS), World Values Surveys (WVS), Eurobarometer surveys, Latinobarometer surveys, Asian- barometer surveys, UK Labour Force Sur- vey (LFS)	Eurobarometer surveys	British Household Panel Survey (BHPS), German Socio-Economic Panel (SOEP), Household, Income and Labour Dynam- ics in Australia (HILDA), Medicine in Australia Balancing Employment and Life (MABEL)
Age Coverage	20 to 80	N/A	varies depending on survey used	15 to 65	sample consists of people aged between 16 and 70
Age Variable	continuous age function, colort indicators	continuous age function	continuous age function, age cohort indi- cators	step function defined for different age groups (10-year intervals)	continuous age function for the mean yearly changes in overall life satisfaction
Life Satisfaction Measure	life satisfaction measure (0=low to 10=high)	Happiness measure, US GSS (3-point scale), GB Eurobarometer surveys (4-point scale)	different psychological well-being measures (mental distress, work satisfaction, life sat- isfaction, depression) depending on survey used	consumption of anti-depressants, mental health issues	life satisfaction measure, BHPS (1 to 7), HILDA (0 to 10), SOEP (0 to 10), MABEL (0 to 10)
Controls	marital status, gender, number of chil- dren, employment status, education, in- come, satisfaction with health, panel du- ration	time, marital status, gender, race, employ- ment status, education	education, marital and work force status, income, cohort effects	education, marital and work force status, gender	none
Functional Relationship	wave function	U-shaped relationship in happiness	U-shaped	inverted U-shaped pattern in the use of an- tidepressants	U-shaped
Turning Points	between age 55 and 70	between age 37 and 41, depending on survey type	around midlife depending on surveyed country	around midlife	BHPS (42.3), HILDA (40.1), SOEP (41.4), MABEL (46.9)
Main Findings	Life satisfaction is wave-shaped. Turn- ing points are explained mainly by sample composition (e.g., unhappy people disap- pear between 55 and 70)	lifetime pattern of happiness, the U- shaped relationship between happiness and age is highlighted in a paragraph	life satisfaction (defined by different psy- chological well-being measures) is U- shaped over the life span for a large num- ber of countries	the inverted U-shaped pattern in mental health problems confirms that subjective well-being is U-shaped in age	Life satisfaction is U-shaped over the life span in all four longitudinal datasets

Study Design/Author, Date	Clark (2007)	Clark and Oswald (2006)	Easterlin (2006)	Easterlin and Sawangfa (2007)	Fischer (2009)
Data Type	longitudinal panel data		pooled cross-sectional data	pooled cross-sectional data	cross-sectional data
Sample Size	136,055	about 100,000	29,000	29,000	44,000
Time Coverage	1991-2005	1991-2004	1973-1994	1973-1994	1997-1999, 1999-2001
Number of Individuals	N/A	N/A	N/A	N/A	44,000
Methodology	pooled OLS, random effects, and panel fixed effects regressions	pooled OLS, random effects, and panel fixed effects regressions	ordered logit regressions	ordered logit regressions	pooled OLS regressions
Country	Great Britain	Great Britain	United States	United States	30 OECD Countries
Survey Type	British Household Panel Survey (BHPS)	British Household Panel Survey (BHPS)	US General Social Surveys (GSS)	US General Social Surveys (GSS)	World Values Surveys (WVS)
Age Coverage	16 to 64	16 to 69	18 to 89	18 to 89	15 to 101
Age Variable	step function defined for different age groups (5-year intervals)	step function defined for different age groups (10-year intervals)	continuous age function	continuous age function	continuous age function
Life Satisfaction Measure	life satisfaction measure (0=low to 7=high scale), general health questionnaire score (0=low to 12=high scale)	life satisfaction measure (0=low to 7=high scale), general health questionnaire score (0=low to 12=high scale)	happiness measure (3-point scale), finan- cial satisfaction (3-point scale), family sat- isfaction (7-point scale), health satisfac- tion (7-point scale), and job satisfaction (4-point scale)	happiness measure (3-point scale), finan- cial satisfaction (3-point scale), family sat- isfaction (7-point scale), health satisfac- tion (7-point scale), and job satisfaction (4-point scale)	life satisfaction measure (1=low to 10=high)
Controls	education, marital and work force status, income, health and cohort effects	education, marital and work force status, income, health	race, gender, education and birth cohort effects	race, gender, education and birth cohort effects	education, marital and work force status, income, cohort effects, political ideology, religion, trust, and social capital
Functional Relationship	U-shaped	U-shaped	inverted U-shaped	inverted U-shaped	wave function
Turning Points	between age 45 and 49	between ages 40 and 49	at age 51	50	first (second) turning point between ages 41 and 46 (80 and 90)
Main Findings	life satisfaction is U-shaped with respect to age, which can be shown by plotting the coefficients of the step function, cohort effects do not fully explain these results	life satisfaction is U-shaped in age, which can be shown by plotting the coefficients of the step function	happiness of a birth cohort rises slightly from age 18 to midlife and declines slightly thereafter	the relationship between mean happiness and age varies by education, year and birth cohort, respectively. This holds for differ- ent satisfaction domains	empirical evidence suggests a wave-shaped relationship between life satisfaction and age

Study Design/Author, Date	Friiters and Beatton (2012)	Galambos et al. (2015)	et al. (2015) Gerstorf et al. (2008) Graham and Ruiz	Graham and Ruiz Pozuelo (2017)	Gwozdz and Sousa-Poza (2010)
Data Type	longitudinal panel data	longitudinal panel data	longitudinal/panel	pooled cross-sectional data	longitudinal panel data
Sample Size	SOEP (176,770), BHPS (153,886), HILDA (75,529)	N/A	10, 162	about 230,000	241,000
Time Coverage	SOEP (1984-2002, former West Ger- many sample), BHPS (1991-2009), HILDA (2001-2008)	7 waves from 1986 to 2010	1984-2005	2005-2014	1994-2006
Number of Individuals	SOEP (about 20,000), BHPS (about 2,500), HILDA (about 13,000)	968 for high school and 574 for university sample, respectively	1,637	N/A	39,000
Methodology	panel fixed effects regressions	hierarchical linear regression models	panel fixed effects and random effects re- gressions	OLS regression models	panel fixed effects regressions
Country	Germany, Great Britain, Australia	Canada (high school seniors; university se- niors; two samples)	Germany	46 Countries from the GWP (Gallup World Poll)	Germany
Survey Type	Gernan Socio-Economic Panel (SOEP), British Household Panel Survey (BHPS), Household Income Labour Dynamics Aus- tralia (HILDA)	Special surveys on Canadian high school seniors and university seniors with chang- ing survey method (mail/telephone)	German Socio-Economic Panel (SOEP)	GWP (Gallup World Poll)	German Socio-Economic Panel (SOEP)
Age Coverage	SOEP (18 to 92), BHPS (18 to 90), HILDA (18 to 92)	18 to 43 and 23 to 37	71 to 101	N/A	16 to 94
Age Variable	continuous age and step function (5-year intervals)	growth curve analysis of age	continuous age and distance to death func- tion	continuous age function	step function with different age groups (5- year intervals)
Life Satisfaction Measure	SOBP (scaled 0-10), BHPS (scaled 1-7), HILDA (scaled 0-10)	happiness (3-point scale)	life satisfaction (11-points scale)	life satisfaction (11-point scale) and stress measure (2-point scale, yes/no variable)	life satisfaction (11-points scale)
Controls	education, marital and work force status, income, health and cohort effects, years in panel	gender, parents' education, grades, self- esteem, marital status, unemployment, self-rated physical health	gender, education, and various interactions with age	marital status, gender, education, employ- ment	marital status, labour force status, number of children, health, education
Functional Relationship	U-shape, flat	inverted U-shaped	linear	U-shaped	wave-shape
Turning Points	first turning point around age 55, second turning point around age 67	early 30s	N/A	varies across countries	first (second) turning point between ages 45 and 49 (70 and 74)
Main Findings	relative to 20-year olds, there is not much effect until around age 55, then a slight increase up to age 67, followed by a sharp decline around age 75	The inverted U-shaped relationship be- tween happiness and age points to the ad- vantage of the use of longitudinal panel data and panel regression models	life satisfaction declines over the life span, with a pronounced decline four years prior to death	Turning point varies with average happi- ness of a country and with the position in the well-being distribution. The U-curve (happiness and age) and the hump-shape (stress and age) are robust	life satisfaction shows a wave-shaped pat- tern over the life span, which disappears when controlling for birth cohort effects

Study Design/Author, Date	Hayo and Seifert (2003)	Kassenboehmer and Haisken-DeNew (2012)	Lang et al. (2011)	Lobos et al. (2016)	McAdams et al. (2012)
Data Type	pooled cross-sectional data	longitudinal panel data	pooled cross-sectional data	cross-sectional data	longitudinal panel data
Sample Size	165,11	116,876	about 100,000	385	N/A
Time Coverage	1991-1995	1994-2006	1997-2006	May 2013, January 2014	1996-2000; 2002-2004
Number of Individuals	N/A	N/A	N/A	385	24,094
Methodology	ordered logit regressions	panel fixed effects regressions	logistic regressions	ordered logit regressions	multilevel modelling to predict the out- come variable
Country	11 Eastern European countries	Germany	England	Maule region in Chile	Great Britain
Survey Type	special survey conducted by the Paul Lazarsfeld Society	German Socio-Economic Panel (SOEP)	annual Health Survey for England (HSE)	special survey on elderly people	British Household Panel Survey (BHPS)
Age Coverage	18 to 77	N/A	aged >15 years	60 to 90	16 to 93
Age Variable	continuous age function	continuous age function	logistic regressions by age group	continuous age function	continuous age function
Life Satisfaction Measure	financial satisfaction	life satisfaction measure (0=low to 10=high)	12-item Generalized Health Questionnaire (GHQ) scores, mental illness diagnoses, and receipt of relevant medication	happiness $(3.$ point scale) calculated from the sum of four items $(7-$ points scale)	eight domains of life satisfaction measures (7-point scale)
Controls	ideology, economic situation five years ago and in five years, gender, income, educa- tion, employment, marital status, religion	marital and employment status, health, in- come, household composition, respondent's survey experience and interviewer fixed ef- fects	ethnicity, income, smoking, social class, education, co-morbidities	employment, gender, partnership, several scales on subjective satisfaction with cir- cumstances	IIONE
Functional Relationship	U-shaped	U-shape, flat	inverted U-shaped relationship in mental illness measures is consistent with the U- shaped pattern in life satisfaction over the life span	inverted U-shaped	wave-shape pattern in life satisfaction, but argumentation mostly about the U-shape up to 75 years
Turning Points	around age 37	N/A	between ages 45 and 54	around age 77.5	first (second) turning point between ages 36 to 44 (72 to 76)
Main Findings	life satisfaction is U-shaped in age. The effect is robust to the inclusion of other socio-demographic and economic variables	the U-shaped pattern of well-being disap- pears once controlling for interviewer and panel duration fixed effects	The prevalence of increased mental issues in midlife is more prominent in low-income households	happiness is U-shaped over the life span. Public policies on older adults in rural areas should account for subjective well- being factors and not only for objective ones	satisfaction in some life domains increases after middle age, while it decreases in oth- ers, life satisfaction may balance out over- all

Study Design/Author, Date	Morgan et al. (2015)	Mroczek and Spiro (2005)	Mroczek and Kolarz (1998)	Piper (2015)	Schwandt (2016)
Data Type	cross-sectional data	longitudinal panel data	cross-sectional data	longitudinal panel data	longitudinal panel data
Sample Size	46,301	8,997	2,727	30,942	132,609
Time Coverage	2012	1978-1999	1995, 1996	BHPS (1996-2000, 2002-2007), wave 11 (2001) excluded	1991-2002
Number of Individuals	N/A	1,927	2,727	9,821	23,161
Methodology	structural regression models	panel fixed effects and random effects re- gression models	OLS regression models	panel fixed effects regression for static and panel GMM regressions for dynamic panel data models	non-parametric approach in the analysis of life satisfaction forecast errors
Country	29 European countries	United States	United States	Great Britain	Germany
Survey Type	European Social Surveys (ESS)	Veteran Affairs (VA) Normative Aging Study (NAS)	Midlife in the United States Survey (MIDUS)	British Household Panel Survey (BHPS)	German Socio-Economic Panel (SOEP)
Age Coverage	20 to 79	40 to 85	25 to 74	16 to 30	17 to 85
Age Variable	continuous age function	continuous age function	continuous age function	step function with different age groups (4- years intervals)	step function for each age
Life Satisfaction Measure	three domains of hedonic well-being and ten domains of eudaimonic well-being	life satisfaction measure	positive/negative affect	life satisfaction (7-point scale)	forecast errors in life satisfaction over a 5- year period
Controls	N/A	N/A	personality, socio-demographic, and con- textual variables	income, job status, marital status, educa- tion, health	gender, East/West Germany dumny, edu- cational status
Functional Relationship	life satisfaction is U- shaped/flat/increasing over the life span	inverted U-shaped	positive affect is U-shaped while negative affect declines linearly with age	U-shaped	wave-shape
Turning Points	around age 50s, but results only hold for high-income countries	around age 65	around age 35	N/A	mid 50s and 70s
Main Findings	life satisfaction is rather flat (negative) over the life span in high-income (low- income) countries. The U-shaped pattern in life satisfaction is conditional on na- tional GDP per capita income levels	life satisfaction increases until age 65 and then declines. The rate of change depends on personal characteristics (e.g., extraver- sion)	personality, contextual and socio- demographic variables, as well as their interactions contribute significantly to the age-affect relationship	life satisfaction is U-shaped in age. Results from the dynamic panel analysis indicate that this finding is a lifecycle effect	young people overestimate, while elderly underestimate future life satisfaction. The U-shape of the age-well-being relationship in midlife is consistent with the hypothesis of unnet aspirations

Study Design/Author, Date	Stone et al. (2010)	Van Landeghem (2008)	Van Landeghem (2012)	Wooden and Li (2014)	Wunder et al. (2013)
Data Type	cross-sectional data	longitudinal panel data	longitudinal panel data	longitudinal panel data	longitudinal panel data
Sample Size	340,847	206,911	N/A	130,211	BHPS (123,656), SOEP (253,044)
Time Coverage	2008	1985-2007	1985-2007	2001-2010	BHPS (1996-2006), SOEP (1986-2007)
Number of Individuals	340,847	32,470	44,960	21,280	BHPS (23,785), SOEP (33,451)
Methodology	weighted least squares regression models	pooled OLS and panel fixed effects regressions	pooled OLS regressions	panel fixed effects regressions	semi-parametric regression model with pe- nalized splines (P-splines)
Country	United States	Gernany	Germany	Australia	Germany, Great Britain
Survey Type	Gallup World Poll (GWP)	German Socio-Economic Panel (SOEP)	German Socio-Economic Panel (SOEP)	Household Income Labour Dynamics Aus- tralia (HILDA)	German Socio-Economic Panel (SOEP), British Household Panel Survey (BHPS)
Age Coverage	18 to 85	aged >17 years	17 to 85	aged >14 years	N/A
Age Variable	step function with different age groups (3- years intervals)	continuous age function, step function with different age groups (4-years inter- vals)	continuous age function	continuous age function	P-splines to estimate the smooth function for age
Life Satisfaction Measure	measures of global (long term) and hedonic (affective) well-being	life satisfaction (11-point scale)	life satisfaction (11-point scale)	life satisfaction (11-points scale)	life satisfaction, BHPS (7-points scale), SOEP (11-points scale)
Controls	gender, partnership, children, employment status	marital status, employment status, cohort effects, income, gender, education	marital status, employment status, cohort effects, income, gender, education	home ownership, equity, location, inter- view mode, labor force status, panel at- trition to control for selectivity bias	N/A
Functional Relationship	U-shaped pattern in positive hedonic and global well-being, inverted U-shaped pat- tern in negative hedonic well-being	U-shape, wave-shape	life satisfaction is U-shaped in age using pooled OLS	third-order polynomial function for age	the semi-parametric approach allows for more flexibility in the well-being-age pat- tern
Turning Points	around the 50s	between ages 42 and 52	42-52	first (second) turning point around age 40 (70)	first (second) turning point around age 48 (78) for BHPS and 52 (65) for SOEP
Main Findings	global and positive hedonic well-being is generally U-shaped over the life span. The inclusion of additional controls that might co-vary with the age-well-being pattern did not change the main findings	U-shape in pooled OLS regressions. No or small positive effect in fixed effects models that control for birth cohort effects	life satisfaction is U-shaped in age using pooled OLS, although other patterns can- not be excluded. The challenge is to disen- tangle age, time, and cohort effects on life satisfaction	the cubic relationship between life satis- faction and age becomes very flat during much of midlife. Findings are still signif- icant when controlling for panel attrition effects	life satisfaction is wave-shaped over the life span

Due to the large variety of methodological approaches, it is not possible to reach a definitive conclusion as to which set-up would be the best choice and how specific methodological issues might affect the results. We therefore used a common empirical framework with a single representative household survey (SOEP) where we could account for cross-sectional and longitudinal variation across survey respondents and a variety of control variables, and tested for different functional relationships between well-being and age.

3 Estimation Methodology

To disentangle the between and within effects of individual-level control variables on subjective well-being, we followed the methodology in Mundlak (1978) and estimated a hybrid version of the correlated random effects (CRE) framework. As shown below, this model framework goes beyond the commonly used fixed effects (FE) regression model by *simultaneously* estimating the between and within effects of time-variant individual-level controls. It is worth mentioning that this methodology even allows the inclusion of time-constant variables that could otherwise not be estimated when using the FE estimation framework.

We start the discussion on the proposed empirical framework using the following linear regression model

$$LS_{it} = \beta + \beta_D \mathbf{D}_{it} + \beta_E \mathbf{E}_{it} + \beta_{ID} \mathbf{I} \mathbf{D}_{it} + \beta_G \mathbf{G}_{rt} + \beta_Z \mathbf{Z}_i + \varepsilon_{it}, \tag{1}$$

where LS is the well-being (or life satisfaction) of individual *i* at time *t*, \mathbf{D}_{it} is a vector of demographic controls (e.g., age, marital and health status), \mathbf{E}_{it} is a vector of economic controls (e.g., personal income, education and employment status), \mathbf{ID}_{it} is a vector of interviewer design controls (e.g., interviewer ID and survey-type fixed effects), \mathbf{G}_{rt} is a set of various time fixed effects that vary across regions and years (e.g., state-year fixed effects), and \mathbf{Z}_i is a vector of time-constant individual characteristics (e.g., gender and native status).

The strict exogeneity assumption between the explanatory variables and the error term ε_{it} determines the estimation methodology of this regression equation. We first discuss the FE regression model for estimating within effects on subjective well-being that are robust to unobserved individual-level heterogeneity. We then contrast this model specification with the CRE framework and show that the latter approach has the advantage of estimating simultaneously between and within effects of explanatory variables.

In the FE estimation framework, the composite error term takes the form $\varepsilon_{it} = \alpha_i + v_{it}$, which results in the following regression equation

$$LS_{it} = \beta + \beta_D^{FE} \mathbf{D}_{it} + \beta_E^{FE} \mathbf{E}_{it} + \beta_{ID}^{FE} \mathbf{I} \mathbf{D}_{it} + \beta_G^{FE} \mathbf{G}_{rt} + \beta_Z \mathbf{Z}_i + \alpha_i + v_{it},$$
(2)

where α_i is the individual-specific (unobserved) heterogeneity that is allowed to be arbitrarily correlated with the set of explanatory variables, e.g., $\mathbb{E}[\alpha_i | \mathbf{D}_{it}, \mathbf{E}_{it}, \mathbf{ID}_{it}, \mathbf{G}_{rt}] \neq 0$. The FE approach effectively controls for time-invariant individual-specific factors that may be correlated with the time-variant explanatory variables. The coefficient vector on the time-variant explanatory variables was identified using the within-variation across individual observations. However, the robustness of the coefficient estimates in the presence of time-constant omitted factors comes at the cost that the effects of any individual-specific time-invariant variables (e.g., gender and ethnic background) are indistinguishable from the unobserved effect α_i , and thus cannot be estimated when using the FE estimation approach. In addition, coefficient estimates of rarely changing variables will be inefficient because of insufficient within-individual variation. For example, the effect of education, marital status, and employment will be hard to assess in an FE framework since these variables change very slowly over time and therefore cannot be fully distinguished from the arbitrary individual-level effect α_i .

In the CRE framework, the relationship between the unobserved effect α_i and the time-variant explanatory variables is explicitly modeled to relax the restrictive zero correlation assumption $\mathbb{E}[\alpha_i | \mathbf{D}_{it}, \mathbf{E}_{it}, \mathbf{ID}_{it}, \mathbf{G}_{rt}] = 0$ within the random effects (RE) framework (Wooldridge, 2010a). Mundlak (1978) showed that the concrete specification of the unobserved effect makes the distinction between the FE and RE approach arbitrary and unnecessary. Following the notational convention in Wooldridge (2010a), we write the composite error in the CRE framework as

$$\varepsilon_{it} = \boldsymbol{\pi}_D \overline{\boldsymbol{D}}_i + \boldsymbol{\pi}_E \overline{\boldsymbol{E}}_i + \boldsymbol{\pi}_{ID} \overline{\boldsymbol{I}} \overline{\boldsymbol{D}}_i + \boldsymbol{\pi}_G \overline{\boldsymbol{G}}_i + a_i + v_{it}, \tag{3}$$

where a bar over a variable refers to time averages of the time-variant explanatory variables calculated on the set of time periods of individual *i* employed in the estimation. In contrast to the standard RE assumption, the unobserved heterogeneity effect α_i is allowed to be correlated with the time-variant explanatory variables according to the relationship $\alpha_i = \pi_D \overline{D}_i + \pi_E \overline{E}_i + \pi_{ID} \overline{ID}_i + \pi_G \overline{G}_i + a_i$, where a_i is now assumed to be uncorrelated with the time-variant explanatory variables in a standard RE estimation framework. Substituting expression (3) into Equation (1) results in the following CRE regression equation, as first proposed by Mundlak (1978)

$$LS_{it} = \beta + \beta_D^{FE} \mathbf{D}_{it} + \beta_E^{FE} \mathbf{E}_{it} + \beta_{ID}^{FE} \mathbf{I} \mathbf{D}_{it} + \beta_G^{FE} \mathbf{G}_{rt} + \beta_Z \mathbf{Z}_i + \pi_D \overline{\mathbf{D}}_i + \pi_E \overline{\mathbf{E}}_i + \pi_{ID} \overline{\mathbf{I}} \overline{\mathbf{D}}_i + \pi_G \overline{\mathbf{G}}_i + a_i + v_{it}.$$
(4)

Extending the basic random effects regression model by including individual-level group means of time-variant explanatory variables effectively controls for arbitrary individual-level heterogeneity. Thus, the RE estimate of the time-variant explanatory variables (e.g., $\beta_D^{FE}, \beta_E^{FE}, \beta_{ID}^{FE}, \beta_G^{FE}$) are identical to the FE estimate in Equation (2).⁶ More importantly, the coefficient estimates of the individual-level group means (e.g., $\pi_D, \pi_E, \pi_{ID}, \pi_G$) refer to the difference of the between and within effects, i.e., $\pi_{\delta} = (\beta_{\delta}^B - \beta_{\delta}^{FE})$ for $\delta = (D, E, ID, G)$.

It is worth mentioning that the formulation in Equation (4) is analogous to a simple regression-based test of the Hausman statistic in comparing the RE and FE estimators (Wooldridge, 2010b). It can be shown that the RE estimate is a matrix-weighted average of the within and between estimator (Baltagi, 2008). Therefore, specifying the hypothesis that π_{δ} equals zero would provide a formal statistical test of the differences in coefficient estimates from the FE and RE approach. Rejection of the null hypothesis that π_{δ} equals zero would suggest that substantial individual-level heterogeneity is present in the data.

Given the fact that the coefficient estimate of the individual-level group means refers to the difference of the within and between estimate, Allison (2009) suggested the following hybrid version in the CRE framework in Equation (4)

$$LS_{it} = \beta + \beta_D^{FE}(\mathbf{D}_{it} - \overline{\mathbf{D}}_i) + \beta_E^{FE}(\mathbf{E}_{it} - \overline{\mathbf{E}}_i) + \beta_{ID}^{FE}(\mathbf{I}\mathbf{D}_{it} - \overline{\mathbf{I}}\overline{\mathbf{D}}_i) + \beta_G^{FE}(\mathbf{G}_{rt} - \overline{\mathbf{G}}_i) + \beta_Z \mathbf{Z}_i + \beta_D^B \overline{\mathbf{D}}_i + \beta_E^B \overline{\mathbf{E}}_i + \beta_{ID}^B \overline{\mathbf{I}}\overline{\mathbf{D}}_i + \beta_G^B \overline{\mathbf{G}}_i + a_i + v_{it}.$$
(5)

Estimating this equation under the standard RE assumption, the coefficient estimates of the time-variant explanatory variables, again, are identical to the FE estimates. The only difference refers to the interpretation of the coefficient

⁶The interested reader is referred to Mundlak (1978) and Wooldridge (2010b) for a technical proof.

estimates of the individual-level group means. Under the CRE model, this coefficient estimate indicates the difference of the within and between estimate, whereas in the hybrid version, this estimate directly refers to the between estimate (Schunck, 2013). Although the FE estimate in the hybrid version remains unbiased irrespective of whether one includes individual-level group means in Equation (5), its inclusion ensures that the coefficient estimate on the time-invariant individual-level controls (e.g., \mathbf{Z}_i) is robust to between-individual-level heterogeneity in the time-variant explanatory variables (Schunck, 2013).

In contrast to the FE approach, the proposed CRE framework provides a number of statistical advantages that allow for more in-depth analysis of key research questions (Bartels, 2015). The hybrid version of the CRE estimation approach can be seen as a unified estimation framework that explicitly separates the within and between estimates in a standard RE framework. For example, the within coefficient estimate regarding age would predict how individual well-being is affected as the particular person becomes older *over* time. This would correspond to a purely individual-specific effect, and one should strictly avoid interpreting this effect for between-individual comparisons. The between-individual effect of age is consistently estimated when focusing on the coefficient estimate associated with the mean age variable. In this case, the coefficient of the mean age variable would provide evidence as to how individual well-being is affected when we compare two individuals of different ages.

Second, it is noteworthy that the CRE framework, in contrast to the FE approach, allows the inclusion of key explanatory variables that do not vary over time (e.g., gender and ethnic background). More importantly, this model framework provides an appropriate statistical tool to evaluate the between estimate of slowly changing variables (e.g., education, marital status, and employment) on individual well-being, where the within-estimate in the FE framework would usually indicate no statistical association due to insufficient within variation of observational units.

Finally, contrasting the within and between estimates in a single regression model provides a statistical framework to test the equality of the within and between estimate of key explanatory variables.

4 Data and Variables

Throughout the empirical analysis, we have used data from the German Socio-Economic Panel (SOEP). It is a representative panel that contains a wide range of survey data on the personal and household level from 1984 to 2016. The survey is conducted in annual waves of around 20,000 individual respondents and 11,000 households per year. The individuals are interviewed repeatedly, making the SOEP one of the longest-running panel studies worldwide. It is one of the most frequently used data sources in happiness research (Wagner et al., 2007). The SOEP was extended in 1990 to include former East Germany. We use data from 1992 onwards to avoid bias caused by the effect of reunification in 1990. Due to this restriction, to missing information on interviewer characteristics in several observations, and to the removal of individuals who took part only once in order to prevent a bias of the within effect in comparison to the between effect, we ended up with a sample of 49,756 individuals and 382,012 observations.

The main questionnaire consists of questions on household and personal characteristics as well as other individual and socio-economic characteristics. The question about individual life satisfaction that we use as our measure of SWB appears in the questionnaires as: "How satisfied are you at present with your life, all things considered? Please respond using the following scale, where '0' indicates not at all satisfied and '10' indicates completely satisfied".

We used the standard correlates found to influence subjective well-being in the literature (e.g., health, marital status, income, number of children, education, employment, and gender).⁷ Furthermore, we controlled for interviewer fixed effects and the experience of individuals within the panel. We therefore followed the suggestion in Kassenboehmer and Haisken-DeNew (2012), pointing out that the U-shaped relationship between subjective well-being and age is sensitive to the inclusion of these variables.

To control for income differences across survey respondents, we used information on monthly net household income (in EUR) reported by the head of the household. Furthermore, we controlled for the employment status of an individual using a dummy indicating whether he or she was employed at the time of the interview. To account for individual health status, we used a measure of self-rated current health status on a five-point Likert scale ranging from *very good*=1 to *bad*=5. For gender, we used a *Male* indicator that takes a value of 1 for male and 0 for female respondents. We also implemented a dummy for ethnicity with respect to country of birth, which takes a value of 1 for individuals born in Germany or German immigrants prior to the year 1949 and 0 otherwise. We did not differentiate among ethnicities other than German since the respective categories would be very small. Referring to the differences in the education of the respondents, we used a set of indicator variables to distinguish among individuals with less than secondary education and those with completed secondary education . To control for marital status, we used five categories distinguishing whether an individual is single, married, separated, divorced, or widowed. The survey experience of individuals was measured by the number of times they participated in the survey. The same holds for interviewer experience.

5 Empirical Results

Tables 7 to 9 present the results on the SWB and age relationship using the proposed Mundlak (1978) estimation approach. In Table 7, the age variable enters in a simple linear form into the regression equation. In Tables 8 and 9, we used a quadratic and cubic functional relationship of the age variable, respectively, to examine a possible non-linear (e.g., Ushaped or wave-shape) pattern of well-being over the life span.

Following the relevant literature, we used a standard set of individual level controls throughout all model specifications: log of net income, employment status, health status, gender, country of origin of the respondent, educational status, marital status, number of children, birth decade FE, panel attrition, survey-type FE, panel individual level FE, and a full set of state-year FE. The inclusion of state-year FE in the regression model captures time-variant unobserved factors across the 16 German federal states, such as differences in labor market policies, infrastructure quality, and public assistance to certain industrial sectors that affect all persons within the same federal state equally. In all three tables, the results of this basic specification are given in column (1). Thereafter, we subsequently introduced survey experience, interviewer gender, interviewer experience, and interviewer FE into the regression model (columns 2 to 5).

The regression in Table 7 column (1) tests a simple linear association between SWB and individual's age. It is worth mentioning that the proposed Mundlak estimation approach clearly distinguishes between within-person and betweenperson effects. Thus, in panel A, the results of the within-person dimension are reported, while panel B refers to the between-person effects. Regarding the basic control variables log net income, employment and health, the estimated

⁷For reviews on the determinants of subjective well-being in the literature, see Diener et al. (1999) and Dolan et al. (2008).

regression coefficients are of the expected signs and statistically significant at conventional significance levels. Furthermore, within and between effects are in line with each other, i.e., sign and significance are the same along the two dimensions. These findings hold throughout all model specifications.

Concerning the effect of age on SWB, the within effect is insignificant, suggesting that age ceteris paribus does not have a separate impact on SWB (see Table 7, column (1), Panel A). In contrast, the between effect shows a positive and highly significant effect of age on SWB (Panel B). Thus, the within and between effects of age on SWB are not consistent with each other.

The specification presented in column (2) examines the sensitivity of the main results to the individual's survey experience. Specifically, we constructed a years-in-panel variable for each respondent that reflects the number of times a respondent appears in the survey. This variable was intended to control for the possibility that respondents spending many years in the survey may become more aware or honest about their level of happiness (Frijters and Beatton, 2012; Kassenboehmer and Haisken-DeNew, 2012). We used a logarithmic specification of the years-in-panel variable to account for the possibility that survey respondents may settle relatively quickly at a reported level of happiness after a few survey rounds. This is in line with Wooden and Li (2014), who found that in their quadratic modelling the turning point was located at the end of the observed time span. The identification of the years-in-panel effect from the individual's age (both monotonically increasing over time) was maintained due to panel exit and entry of survey respondents over time. The estimates in column (2) indicate that longer survey experience does indeed lead to lower reported SWB, both in the within and the between dimension. Reassuringly, the main results regarding the age and SWB relationship are not sensitive to the inclusion of the individual's survey experience.

In columns (3) and (4), we followed the suggestions in Kassenboehmer and Haisken-DeNew (2012) and controlled for the possible influence of interviewer gender and experience (measured as logarithm of the number of years of experience the interviewer has in surveying individuals in the German SOEP). The hypothesis was that male and experienced interviewers may be regarded as trustworthy and thus cause survey respondents to reveal their true level of happiness. The findings in column (4) indicate that both the interviewer's gender and the interviewer's experience have a significant negative impact on the reported SWB. This finding applies to the within as well as to the between effect. Again, the coefficients of the age variables are not sensitive to this model specification.

Finally, we included a full set of interviewer ID FE to control for arbitrary unobserved interviewer heterogeneity that might affect the respondent's survey behavior (column 5).⁸ While the control variables remained largely unchanged, with exception of the between effect of interviewer experience, which now became insignificant, the coefficient of the within effect of age changed substantially. The coefficient increased from 0.0814 to 0.3487 and became highly statistically significant at conventional significance levels, suggesting that SWB increases linearly over the life span. This now mirrored the findings for the between dimension. Note that the within R^2 increased by 1.4 percentage points to 10.2% and the between R^2 by 5.5 percentage points to 45.9%. Figure 1 shows the absolute and marginal within and between effects of age on SWB based on the results in Table 7. As the figure illustrates, the coefficients of the within effect were more sensitive to the introduction of the interviewer FE than those of the between effect.

In Table 8, we tested for a possible quadratic relationship between SWB and age. Interestingly all control variables kept

 $^{^{8}}$ The interviewer gender variable dropped out because it was captured by the interviewer FE.

their sign and significance level, and even the sizes of the coefficients were similar compared to the coefficient estimates reported in Table 7. The same observation held for the different R^2 s, which remained largely unchanged.

However, the findings concerning the relation between age and SWB changed substantially. All specifications indicate a quadratic relation between SWB and age. Yet while the results of the between dimension indicate the well-known U-shaped relation, the relation in the within dimension turns out to be inverse U-shaped. This finding holds throughout all model specifications. As Figure 2 illustrates, the findings concerning the within effect changed substantially with the inclusion of additional control variables. While the absolute within effect in column (1) is negative for all ages, in the remaining specifications, it becomes positive (columns 2 to 5). In addition, the findings in columns (2) to (4) indicate that there is a turning point around the age of 80, the specifications in columns (1), (2) and (5) show no turning point within the sample. For the latter specifications, other functional forms might be suitable as well, e.g., linear and/or a logarithmic specification. On the other hand, the between effect of age on SWB is rather uniform throughout the different model specifications. The absolute effect is negative until the age of 60 (columns 1 to 4) and 70 (column 5), respectively. The turning point is between the age of 30 and 35.

Since we analyzed the same individuals for both the within and between dimensions, the different results for the within and between effects of age on SWB are not justifiable either theoretically or econometrically. This contradiction is therefore an indication of a functional misspecification regarding the relationship between SWB and age.

Finally, Table 9 provides regression coefficients using a third-order polynomial specification in the age variable (i.e., a cubic specification) and its association with SWB. Again, the estimated sign, magnitude, and significance levels of the standard control variables remained largely unchanged throughout all model specifications, although the R^2 s in column (5) are the highest of all model specifications and differ only in the third decimal place.

The cubic relation between SWB and age is affirmed in all model specifications and both for the within and between effect. The coefficients for the between effect of age are highly significant and very robust to the inclusion of different control variables. The size of the coefficients changed only very slightly among the columns (1) to (5). For the within effect, the regression coefficients associated with age squared and age cubed, respectively, remained unchanged in sign and significance and changed only slightly in size. Only the coefficient of the linear age term increased with the introduction of respondent and interviewer control variables from -1.0870 in column (1) to -0.7053 in column (5). Figure 3 illustrates the change of the regression coefficients associated with the age variable throughout all model specifications. Both the function of the absolute and the marginal effect of age on SWB is shifted upwards if the regression equation accounts for additional individual-level controls. This has two distinct consequences. First, with the introduction of the interviewer FE (column 5), the absolute within effect of age turns positive for individuals older than 40 years. In the previous specifications (columns 1 to 4), the absolute within age effect remains always negative, irrespective of an individual's age. Second, the upward shift of the curve of the marginal within effect of age on SWB substantially affects the estimated turning points. While the two turning points of the baseline specification (column 1) were at the ages of 35.4 and 64.2, respectively, the estimated turning points in column (5) including the interviewer FE are at ages 17.8 and 81.6, respectively. Comparing these latter findings to those of the between effect, there is a qualitatively significant difference. The two turning points of the between effects are at ages 35.2 and 88.8 (column 5). Thus, the marginal within effect of age on SWB is positive for a duration of 63.5 years, starting from age 17.8 until the age of 81.6 years. In contrast, the marginal between effect of age on SWB is positive only for 53.6 years, starting from age 35.2 until the age of 88.8 years.

To conclude, what have we learned from the regression results as a whole? Regarding the relationship between age and SWB, the regression results clearly show that the obtained findings are mainly driven by the chosen functional specification of the age variable (i.e., linear, quadratic, or cubic) and whether the identification of the age effect employs the within or between variation in the data. Furthermore, the comparison of the three tested functional forms indicates that only the cubic relation between age and SWB leads to robust and consistent results. In addition, the results show that the within estimates of age on SWB are sensitive to the inclusion of additional individual-level controls. Finally, the recently discussed control variables survey experience and interviewer characteristics indeed have a strong impact on reported SWB measures. Throughout all specifications, the regression coefficient associated with survey experience is negative and highly statistically significant at conventional significance levels. This finding suggests that longer survey experience leads to lower reported SWB measures. In addition, reported SWB measures are lower for survey respondents if the interviewer is male and/or has extensive experience surveying respondents. However, these controls have no significant impact on the coefficients associated with the age variables and the reported R^2 s. In contrast, the introduction of the interviewer FE has a strong effect on the within coefficient associated with the linear age variable and the corresponding R^2 s. While the coefficients of all other age variables remain almost unchanged (both for the within and between estimates), the coefficient of the linear age variable in the within dimension increases in all model specifications, as shown in column (5) of Tables 7 to 9, respectively. Moreover, the model's fit is significantly higher for the cubic functional relationship of the age variable compared to the linear and quadratic specifications.

6 Conclusion

This paper contributes to the extensive but conflicting literature on the evolution of individual life satisfaction across the life span. The empirical findings on the topic are relatively mixed. While the majority of studies report a U-shaped relationship between well-being and age, other studies have found that the association is flat, inverted U-shaped, or that it follows a wave-like pattern. A major caveat of the literature is the lack of comparability among the existing studies, as they use different data sets, econometric methods, control variables, and – based on their assumptions – functional specifications of the age variable used in the regression equations. The large between-study heterogeneity prevents any final conclusions from being drawn about the lifecycle pattern of individual life satisfaction. In essence, the heterogeneous results found on the topic could either reflect a true relationship or simply be a statistical artefact caused by differences in study design.

We use a correlated random effects (CRE) econometric framework to address the identification problem between SWB and age. Based on longitudinal data from the German Socio-Economic Panel (SOEP) during the period 1992 to 2016, we estimate a hybrid version of the CRE model to simultaneously estimate the within and between effects of age on reported SWB measures. The proposed estimation framework is designed to provide evidence as to whether the inconsistent results found in the literature regarding the relationship between well-being and age (e.g., U-shaped versus flat relationship) might be caused, among other factors, by the type of data set employed (i.e., cross-sectional versus longitudinal panel data).

We investigate the (possible nonlinear) lifecycle pattern of individual life satisfaction using higher-order polynomials of the age variable. The results show that a cubic functional specification in the age variable provides a good approximation of

the pattern of well-being over the entire life span. Furthermore, the regression coefficients associated with the age variable based on the cubic specification are rather robust to the inclusion of additional individual level controls. In contrast to the linear and quadratic specification of age, the cubic functional form leads to consistent within and between effects of age on SWB. A noteworthy result is that the often cited U-shaped association between SWB and age reported in the majority of studies is only present in the between dimension, while the within effect reveals an inverted U-shaped relationship. We take this finding as evidence that a quadratic specification is too restrictive to provide consistent within and between effects of age on well-being.

Furthermore, we test the robustness of the relationship between well-being and age to the inclusion of a large battery of additional individual level controls. We start with a basic set of sociological, demographic and economic control variables such as gender, immigration status, income, employment, educational, and marital status, and among other factors. We then test the sensitivity of the main findings to the inclusion of recently discussed interviewer factors that might affect the respondent's survey behavior such as interviewer gender and interviewer survey experience. Indeed, our results indicate that respondents interviewed by a male and/or more experienced interviewer report on average lower SWB. This finding is consistent with the idea that survey respondents might be more honest about their true level of SWB if interviewed by a male and/or more experienced interviewer. Additionally, accounting for all kind of time-invariant interviewer characteristics by including interviewer FE into the regression equation improves the model's fit significantly. Regarding the estimated regression coefficients associated with the age variables, the wave-like pattern in well-being remains qualitatively unaffected by the inclusion of individual-level controls throughout all model specifications.

References

Allison, P. D. (2009). Fixed Effects Regression Models. Thousand Oaks, CA: Sage.

- Baetschmann, G. (2014). Heterogeneity in the Relationship Between Happiness and Age: Evidence From the German Socio-Economic Panel. German Economic Review, 15:393–410.
- Baltagi, B. H. (2008). Econometric Analysis of Panel Data. Wiley, UK: Chichester, West Sussex, fourth edition.
- Bartels, B. L. (2015). Beyond 'Fixed versus Random Effects': A Framework for Improving Substantive and Statistical Analysis of Panel, TSCS, and Multilevel Data. In Franzese, R. J., editor, *Quantitative Research in Political Science*, volume 4. SAGE.
- Blanchflower, D. and Oswald, A. (2004). Well-Being Over Time in Britain and the USA. Journal of Public Economics, 88:1359–1386.
- Blanchflower, D. and Oswald, A. (2008). Is Well-Being U-Shaped Over the Life Cycle? Social Sience & Medicine, 66:1733–1749.
- Blanchflower, D. and Oswald, A. (2016). Antidepressants and Age: A New Form of Evidence for U-Shaped Well-Being Through Life. Journal of Economic Behavior & Organization, 127:46–58.
- Brassen, S., Gamer, M., Peters, J., Gluth, S., and Bückel, C. (2012). Don't Look Back in Anger! Responsiveness to Missed Chances in Successful and Nonsuccessful Aging. *Science*, 336:612–614.
- Charles, S. and Carstensen, L. (2009). Socioemotional Selectivity Theory. In Reis, H. and Spencer, S., editors, *Encyclopedia of human relationships*, pages 1578–1581. Sage Publications.
- Cheng, T. C., Powdthavee, N., and Oswald, A. J. (2017). Longitudinal Evidence for a Midlife Nadir in Human Well-Being: Results from Four Data Sets. *Economic Journal*, 127:126–142.
- Clark, A. (2007). Born to Be Mild? Cohort Effects Don't (Fully) Explain Why Well-Being is U-Shaped in Age. *IZA Discussion* Paper No. 3170, pages 1–29.
- Clark, A. and Oswald, A. (2006). The Curved Relationship Between Subjective Well-Being and Age. Paris-Jourdan Sciences Economiques Working Paper, 29:1–12.
- Diener, E. and Lucas, R. (1999). Personality and subjective well-being. In Kahneman, D., Diener, E., and Schwarz, N., editors, Well-Being: The Foundations of Hedonic Psychology, pages 213–229. SAGE, New York.
- Diener, E., Suh, E., Lucas, R., and Smith, H. (1999). Subjective Well-Being: Three Decades of Progress. Psychological Bulletin, pages 276–302.
- Dolan, P., Peasgood, T., and White, M. (2008). Do We Really Know What Makes Us Happy? A Review of the Economic Literature on the Factors Associated with Subjective Well-Being. *Journal of Economic Psychology*, 29:94–122.
- Easterlin, R. A. (2006). Life Cycle Happiness and its Sources: Intersections of Psychology, Economics, and Demography. *Journal of Economic Psychology*, 27:463–482.
- Easterlin, R. A. and Sawangfa, O. (2007). Happiness and Domain Satisfaction: Theory and Evidence. IZA Discussion Paper No. 2584, pages 1–35.
- Fischer, J. A. V. (2009). Happiness and Age Cycles Return to Start... ? OECD Social, Employment and Migration Working Papers, No. 99, OECD Publishing, Paris.

- Fisher, I. (1930). The Theory of Interest. Macmillan, New York.
- Frederick, S., Loewenstein, G., and O'Donoghue, T. (2002). Time Discounting and Time Preference: A Critical Review. Journal of Economic Literature, 40:351–401.
- Frey, B. S. and Stutzer, A. (2002). Happiness and Economics: How Does the Economy and Institutions Affect Well-Being. Princeton University Press, Princeton, New Jersey.
- Frijters, P. and Beatton, T. (2012). The Mystery of the U-Shaped Relationship Between Happiness and Age. Journal of Economic Behavior & Organization, 82(2):525–542.
- Fukuda, K. (2013). A Happiness Study Using Age-Period-Cohort Framework. Journal of Happiness Studies, 14:135–153.
- Galambos, N. L., Fang, S., Krahn, H. J., Johnson, M. D., and Lachman, M. E. (2015). Up, Not Down: The Age Curve in Happiness From Early Adulthood to Midlife in Two Longitudinal Studies. *Developmental Psychology*, 51:1664–1671.
- Gerstorf, D., Ram, N., Estabrook, R., Schupp, J., Wagner, G. G., and Lindenberger, U. (2008). Life Satisfaction Shows Terminal Decline in Old Age: Longitudinal Evidence From the German Socio-Economic Panel Study (SOEP). Developmental Psychology, 44:1148–1159.
- Graham, C. and Ruiz Pozuelo, J. (2017). Happiness, Stress, and Age: How the U Curve Varies Across People and Places. *Journal* of *Population Economics*, 30:225–264.
- Gwozdz, W. and Sousa-Poza, A. (2010). Ageing, Health and Life Satisfaction for the Oldest Old: An Analysis for Germany. Social Indicators Research, 97:397–417.
- Hayo, B. and Seifert, W. (2003). Life Cycle Happiness and its Sources: Intersections of Psychology, Economics, and Demography. Journal of Economic Psychology, 24:329–348.
- Kahneman, D. (1999). Objective happiness. In Kahneman, D., Diener, E., and Schwarz, N., editors, Well-Being: The Foundations of Hedonic Psychology, pages 3–25. SAGE, New York.
- Kahneman, D., Wakker, P., and Sarin, R. (1997). Back to Bentham? Explorations of Experienced Utility. Quaterly Journal of Economics, 112:375–405.
- Kassenboehmer, S. C. and Haisken-DeNew, J. P. (2012). Heresy or Enlightenment? The Well-Being Age U-Shaped Effect is Flat. *Economics Letters*, 117:235–238.
- Lang, I., Llewellyn, D., Hubbard, R., Langa, K., and Melzer, D. (2011). Income and the Midlife Peak in Common Mental Disorder Prevalence. *Psychological Medicine*, 41:1365–1372.
- Lobos, G., Grunert, K. G., Bustamante, M., and Schnettler, B. (2016). With Health and Good Food, Great Life! Gender Differences and Happiness in Chilean Rural Older Adults. *Social Indicators Research*, 127:865–885.
- López Ulloa, B. F., Møller, V., and Sousa-Poza, A. (2013). How Does Subjective Well-Being Evolve with Age? A Literature Review. *Population Ageing*, 6:227–246.
- Lucas, R. (2007). Adaptation and the Set Point Model of Subjective Well-Being. Current Directions in Psychological Science, 16:75–79.

- McAdams, K. K., Lucas, R. E., and Donnellan, M. B. (2012). The Role of Domain Satisfaction in Explaining the Paradoxical Association Between Life Satisfaction and Age. *Social Indicators Research*, 109:295–303.
- Modigliani, F. and Brumberg, F. (1954). Utility Analysis and the Consumption Function: An Interpretation of Cross-Sectional Data. In Kurihara, K., editor, *Post-Keynesian Economics*. Rutgers University Press.
- Morgan, J., Robinson, O., and Thompson, T. (2015). Happiness and Age in European Adults: The Moderating Role of Gross Domestic Product per Capita. *Psychology and Aging*, 30:544–551.
- Mroczek, D. and Kolarz, C. (1998). The Effect of Age on Positive and Negative Effect: A Developmental Perspective on Happiness. Journal of Personality and Social Psychology, 75:1333–1349.
- Mroczek, D. K. and Spiro, A. (2005). Change in Life Satisfaction During Adulthood: Findings From the Veterans Affairs Normative Aging Study. *Journal of Personality and Social Psychology*, 88:189–202.
- Mundlak, Y. (1978). On the Pooling of Time Series and Cross Section Data. Econometrica, 46:69–85.
- Munsey, C. (2007). Accentuating the Positive Why Older People are Happier. Monitor on Psychology, 38:17.
- Piper, A. T. (2015). Sliding Down the U-Shape? A Dynamic Panel Investigation of the Age-Well-Being Relationship, Focusing on Young Adults. Social Science and Medicine, 143:54–61.
- Ramsey, F. P. (1928). A Mathematical Theory of Saving. Economic Journal, 38:543-559.
- Schunck, R. (2013). Within and Between Estimates in Random-Effects Models: Advantages and Drawbacks of Correlated Random Effects and Hybrid Models. *The Stata Journal*, 13:65–76.
- Schwandt, H. (2016). Unmet Aspirations as an Explanation for the Age U-Shape in Wellbeing. Journal of Economic Behavior & Organization, 122:75–87.
- Small, B., Frantiglioni, L., v.Strauss, E., and Beckmann, L. (2003). Terminal Decline and Cognitive Performance in Very Old Age: Does Cause of Death Matter? *Psychology and Aging*, 18:193–202.
- Small, B. J. and Beckmann, L. (1997). Cognitive Correlates of Mortality: Evidence From a Population-Based Sample of Very Old Adults. *Psychology and Aging*, 12:309–313.
- Stone, A. A., Schwartz, J. E., Broderick, J. E., and Deaton, A. (2010). A Snapshot of the Age Distribution of Psychological Well-Being in the United States. *Proceedings of the National Academy of Sciences*, 107:9985–9990.
- Van Landeghem, B. G. M. (2008). Human Well-Being over the Life Cycle: Longitudinal Evidence From a 20-Year Panel. LICOS Discussion Paper Series, Discussion Paper 213/2008, Katholieke Universiteit Leuven.
- Van Landeghem, B. G. M. (2012). A Test for the Convexity of Human Well-Being Over the Life Cycle: Longitudinal Evidence from a 20-Year Panel. Journal of Economic Behavior & Organization, 81:571–582.
- Wagner, G. G., Frick, J., and Schupp, J. R. (2007). The German Socio-Economic Panel Study (SOEP): Scope, Evolution and Enhancement. Schmollers Jahrbuch: Journal of Applied Social Sience Studies, 127:139–169.
- Weiss, A., King, J. E., Inoue-Murayama, M., Matsuzawa, T., and Oswald, A. J. (2012). Evidence for a Midlife Crisis in Great Apes Consistent with the U-Shape in Human Well-Being. *Proceedings of the National Academy of Sciences*, 109:19949–19952.

- Wooden, M. and Li, N. (2014). Panel Conditioning and Subjective Well-Being. Social Indicators Research, 117:235–255.
- Wooldridge, J. M. (2010a). Correlated Random Effects Models With Unbalanced Panels. mimeo.
- Wooldridge, J. M. (2010b). *Econometric Analysis of Cross Section and Panel Data*. The MIT Press, Cambridge, Massachusetts, second edition.
- Wunder, C., Wiencierz, A., Schwarze, J., and Küchenhoff, H. (2013). Well-Being Over the Life Span: Semiparametric Evidence From British and German Longitudinal Data. *The Review of Economics and Statistics*, 59:154–167.

A Regression Tables

	-		· /	
(1)	(2)	(3)	(4)	(5)
Level Controls				
-0.0384	0.0838	0.0851	0.0814	0.3487***
(0.1252)	(0.1252)	(0.1252)	(0.1253)	(0.1291)
0.2936***	0.2952^{***}	0.2955***	0.2960***	0.2990^{***}
(0.0103)	(0.0102)	(0.0102)	(0.0102)	(0.0102)
0.0481***	0.0563***	0.0563***	0.0562***	0.0549^{***}
(0.0092)	(0.0092)	(0.0092)	(0.0092)	(0.0091)
-0.4850***	-0.4829***	-0.4829***	-0.4828***	-0.4793***
(0.0046)	(0.0047)	(0.0047)	(0.0047)	(0.0046)
	-0.1176***	-0.1185***	-0.1042***	-0.1130***
	(0.0087)	(0.0087)	(0.0090)	(0.0097)
		-0.0255**	-0.0246**	
		(0.0116)	(0.0116)	
			-0.0325***	-0.0340***
			(0.0055)	(0.0120)
l-Level Controls				· · · ·
0.1170***	0.1179***	0.1179***	0.1178***	0.1012***
(0.0171)	(0.0170)	(0.0170)	(0.0170)	(0.0166)
0.5137***	0.5135***	0.5138***	0.5148***	0.5168***
(0.0126)	(0.0126)	(0.0126)	(0.0126)	(0.0128)
0.0528***	0.0551***	0.0552***	0.0546***	0.0327**
(0.0168)	(0.0168)	(0.0168)	(0.0168)	(0.0164)
	· · · · ·	· · · · ·	· · · · ·	-0.9549***
(0.0081)	(0.0081)		(0.0081)	(0.0081)
(*****)	· · · ·	· · · · ·	· · · · ·	-0.0262***
				(0.0094)
	(0.000)	· · · · ·	· · · · ·	(0.000-)
		(0.0111)	· · · · ·	0.0049
				(0.0332)
lividual Loval Controls			(0.0031)	(0.0332)
	-0.1543***	-0.1542***	-0.1542***	-0.1437***
				(0.0094)
				-0.0106
				(0.0152)
	· /			382,012
	· · · · · · · · · · · · · · · · · · ·	,	'	49,756
				0.102
				0.327
				0.459
				Yes
Yes	Yes	Yes	Yes	Yes
	res	res	res	res
	(1) Level Controls -0.0384 (0.1252) 0.2936^{***} (0.0103) 0.0481^{***} (0.0092) -0.4850^{***} (0.0046) (0.0046) (0.0171) 0.5137^{***} (0.0171) 0.5137^{***} (0.0168) -0.9662^{***} (0.0081) (0.0081) (0.0081) (0.0081) (0.0081) (0.00875) (0.291) (0.0150) $(0.0150$	$\begin{tabular}{ c c c c c c } \hline (1) & (2) \\ \hline (2) & (0.0384 & 0.0838 & (0.1252) & (0.1252) & (0.1252) & (0.033) & (0.0102) & (0.0481*** & 0.0563*** & (0.0092) & (0.0092) & (0.0092) & (0.0092) & (0.0092) & (0.0092) & (0.0092) & (0.0046) & (0.0047) & (0.0047) & (0.0046) & (0.0047) & (0.0047) & (0.0087) \\ \hline (1) & (0.016) & (0.0160) & (0.0170) & (0.5137*** & 0.5135*** & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0126) & (0.0051) & (0.0051) & (0.0081) & (0.0081) & (0.0081) & (0.0085) \\ \hline \\ \hline & -0.1537*** & -0.1543*** & (0.0085) & (0.0085) & (0.0085) \\ \hline \\ \hline & -0.1537*** & -0.1543*** & (0.0097) & (0.0097) & (0.0097) & (0.0097) & (0.0097) & (0.0085) \\ \hline \\ \hline & -0.1537*** & -0.1543*** & (0.0088) & (0.0151) & (0.0151) & (0.0151) & (0.0151) & (0.0151) & (0.0151) & (0.0151) & (0.0875) & (0.0883) & (0.291) & (0$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Level Controls

 Table 7: Within and Between Effects of Age on SWB – Linear Age Function

Notes: The dependent variable is self-reported individual well-being, measured from Low = 0 to High = 10. In Net Income is the log of the monthly household net income (in EUR). Employed refers to an indicator variable that takes the value 1 for individuals who are currently employed. Health Status is the current self-rated health status, measured as very good=1 to bad=5. In Survey Experience is the log of the number of years a particular individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for period of years a particular individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for individuals born in Germany or German immigrants prior to the year 1949. State-Year Fixed Effects refer to a set of Federal-State-Year Time Fixed Effects. Education Status refer to a set of indicator variables for individuals with an educational status lower than secondary school or with completed secondary school educatio n. Marital Status refer to a set of indicator variables for individuals being single, married, separated, or divorced. No. of Children refers to the number of children in the household. Birth Decade Fixed Effects refer to a set of indicator variables for each birth decade in the survey. Constant term included but not shown. Panel Attrition Control refers to an indicator variable which is 1 if the individual does not participate in the next wave of the survey. Survey-Type Fixed Effects refer to a set of indicator variables according to the unique Interviewer), verbal and/or written, with/without accompanying person but with interpreter). Interviewer ID Fixed Effects refer to a set of indicator variables according to the unique Interviewer ID number.

Standard errors, clustered at the individual level, are shown in parentheses.

*: Significant at the 10% level. **: Significant at the 5% level. ***: Significant at the 1% level.

		Self-Reported Individual W		÷ /	
	(1)	(2)	(3)	(4)	(5)
Panel A: Within Effects of Individual	l-Level Controls				
Age	0.1043	0.2429*	0.2451*	0.2388*	0.5033***
0	(0.1277)	(0.1277)	(0.1276)	(0.1278)	(0.1318)
Age squared	-0.0136***	-0.0149***	-0.0150***	-0.0148***	-0.0145***
0.1	(0.0028)	(0.0028)	(0.0027)	(0.0028)	(0.0028)
n Net Income	0.2885***	0.2896***	0.2898***	0.2904***	0.2937***
	(0.0103)	(0.0103)	(0.0103)	(0.0103)	(0.0102)
Employed	0.0369***	0.0442***	0.0441***	0.0442***	0.0435***
Employed	(0.0094)	(0.0094)	(0.0094)	(0.0094)	(0.0093)
Health Status	-0.4845***	-0.4823***	-0.4823***	-0.4822***	-0.4787***
lealth Status	(0.0046)	(0.0047)	(0.0047)	(0.0047)	(0.0046)
C F	(0.0046)	· · · ·		()	-0.1162***
n Survey Experience		-0.1200***	-0.1209***	-0.1069***	
		(0.0087)	(0.0087)	(0.0090)	(0.0097)
nterviewer Gender			-0.0265**	-0.0257**	
			(0.0116)	(0.0116)	
n Interviewer Experience				-0.0318***	-0.0326***
				(0.0055)	(0.0120)
Panel B: Between Effects of Individua					
Age	-0.2311***	-0.2350***	-0.2350***	-0.2362***	-0.2593***
	(0.0353)	(0.0353)	(0.0353)	(0.0353)	(0.0348)
Age squared	0.0359***	0.0364^{***}	0.0364^{***}	0.0365^{***}	0.0373***
	(0.0032)	(0.0032)	(0.0032)	(0.0032)	(0.0032)
n Net Income	0.5097***	0.5094***	0.5097***	0.5108***	0.5127^{***}
	(0.0126)	(0.0126)	(0.0126)	(0.0126)	(0.0127)
Employed	0.1016***	0.1047***	0.1048***	0.1043***	0.0831***
	(0.0175)	(0.0175)	(0.0175)	(0.0175)	(0.0171)
Health Status	-0.9566***	-0.9557***	-0.9559***	-0.9559***	-0.9450***
	(0.0082)	(0.0082)	(0.0082)	(0.0082)	(0.0081)
n Survey Experience	(0.000_)	-0.0552***	-0.0553***	-0.0448***	-0.0290***
n Survey Experience		(0.0085)	(0.0085)	(0.0089)	(0.0094)
nterviewer Gender		(0.0000)	-0.0179	-0.0185*	(0.0034)
nterviewer Gender			(0.0110)	(0.0110)	
			(0.0110)	-0.0338***	0.0011
n Interviewer Experience					0.0011
Panel C: Time-Constant Effects of In	dividual-Level Controls			(0.0081)	(0.0331)
Male	-0.1610***	-0.1618***	-0.1617***	-0.1618***	-0.1514***
	(0.0098)	(0.0098)	(0.0098)	(0.0098)	(0.0094)
German-Born	-0.0389***	-0.0271*	-0.0268*	-0.0276*	-0.0287*
	(0.0151)	(0.0152)	(0.0152)	(0.0152)	(0.0152)
Observations	382,012	382,012	382,012	382,012	382,012
Number of Individuals	49,756	49,756	49,756	49,756	49,756
R^2 Within	49,756 0.0876	49,756 0.0885	49,756	49,756	49,750 0.102
R^2 Overall	0.0876	0.292	0.292	0.292	0.328
R^2 Overall R^2 Between	0.291 0.406	0.292 0.406	0.292 0.406	0.292 0.406	0.328 0.461
ndividual Fixed Effects	Yes	Yes	Yes	Yes	Yes
State-Year Fixed Effects	Yes	Yes	Yes	Yes	Yes
Educational Status	Yes	Yes	Yes	Yes	Yes
Marital Status	Yes	Yes	Yes	Yes	Yes
Number of Children	Yes	Yes	Yes	Yes	Yes
Birth Decade Fixed Effects	Yes	Yes	Yes	Yes	Yes
Panel Attrition Control	Yes	Yes	Yes	Yes	Yes
Survey-Type Fixed Effects	Yes	Yes	Yes	Yes	Yes
Interviewer ID Fixed Effects	No	No	No	No	Yes

Table 8: Within and Between Effects of Age on SWB – Quadratic Age Function

Notes: The dependent variable is self-reported individual well-being, measured from Low = 0 to High = 10. In Net Income is the log of the monthly household net income (in EUR). Employed refers to an indicator variable that takes the value 1 for individuals who are currently employed. Health Status is the current self-rated health status, measured as very good=1 to bad=5. In Survey Experience is the log of the number of years a particular individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for individuals appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 interviewer is male. In Interviewer Experience is the log of the number of years a particular interviewer conducts interviews in the SOEP. Male takes the value 1 for males. German-Born takes the value 1 for individuals born in Germany or German immigrants prior to the year 1949. State-Year Fixed Effects refer to a set of Federal-State-Year Time Fixed Effects. Education Status refer to a set of indicator variables for individuals with educational status less than high school or completed high school education. Marital Status refer to a set of indicator variables for individuals being single, married, separated, or divorced. No. of Children refers to the number of children in the household. Birth Decade Fixed Effects refer to a set of indicator variables for each birth decade in the survey. Constant term included but not shown. Panel Attrition Control refers to an indicator variable which is 1 if the individual does not participate in the next wave of the survey. Survey-Type Fixed Effects refer to a set of indicator variables with respect to the type of survey (e.g., phone, self-filled (with/without interviewer), verbal and/or written, with/without accompanying person but with interpreter). Interviewer ID Fixed Effects refer to a set of indicator variables according to the unique Interviewer ID number. Standard errors, clustered at the individual level, are shown i

*: Significant at the 10% level. **: Significant at the 5% level. ***: Significant at the 1% level.

	Dependent Variable	e: Self-Reported Individual We	ell-Being (from $Low = 0$ to $Low = 0$	High = 10)	
	(1)	(2)	(3)	(4)	(5)
Panel A: Within Effects of Individual		0.0000###	0.0000848	0.0000444	
Age	-1.0870***	-0.9309***	-0.9286***	-0.9366***	-0.7053***
	(0.1483)	(0.1482)	(0.1481)	(0.1483)	(0.1518)
Age squared	0.2378***	0.2320***	0.2319***	0.2325***	0.2404***
	(0.0173)	(0.0173)	(0.0173)	(0.0173)	(0.0172)
Age cubed	-0.0159***	-0.0156***	-0.0156***	-0.0156***	-0.0161***
	(0.0011)	(0.0011)	(0.0011)	(0.0011)	(0.0011)
In Net Income	0.2862***	0.2873***	0.2876***	0.2882***	0.2914***
	(0.0103)	(0.0103)	(0.0103)	(0.0103)	(0.0102)
Employed	0.0652***	0.0718***	0.0717***	0.0718***	0.0716***
	(0.0094)	(0.0094)	(0.0094)	(0.0094)	(0.0093)
Health Status	-0.4822***	-0.4801***	-0.4801***	-0.4801***	-0.4765***
	(0.0046)	(0.0046)	(0.0046)	(0.0046)	(0.0046)
n Survey Experience		-0.1165***	-0.1174***	-0.1032***	-0.1123***
		(0.0087)	(0.0087)	(0.0090)	(0.0097)
Interviewer Gender			-0.0263**	-0.0255**	
			(0.0116)	(0.0116)	
In Interviewer Experience				-0.0322***	-0.0340***
				(0.0055)	(0.0120)
Panel B: Between Effects of Individua					
Age	-1.0365***	-1.0768***	-1.0766^{***}	-1.0718***	-1.0721***
	(0.0983)	(0.0986)	(0.0986)	(0.0985)	(0.0965)
Age squared	0.2093***	0.2176^{***}	0.2175^{***}	0.2163***	0.2123***
	(0.0202)	(0.0202)	(0.0202)	(0.0202)	(0.0197)
Age cubed	-0.0112***	-0.0118***	-0.0117***	-0.0117***	-0.0114***
	(0.0013)	(0.0013)	(0.0013)	(0.0013)	(0.0013)
In Net Income	0.4997***	0.4990***	0.4993***	0.5004***	0.5026***
	(0.0126)	(0.0126)	(0.0126)	(0.0126)	(0.0128)
Employed	0.1291***	0.1338***	0.1338***	0.1332***	0.1111***
	(0.0178)	(0.0178)	(0.0178)	(0.0178)	(0.0173)
Health Status	-0.9559***	-0.9549***	-0.9551***	-0.9551***	-0.9440***
	(0.0081)	(0.0081)	(0.0082)	(0.0081)	(0.0081)
In Survey Experience		-0.0601***	-0.0603***	-0.0501***	-0.0349***
		(0.0085)	(0.0085)	(0.0089)	(0.0094)
Interviewer Gender			-0.0178	-0.0183*	
			(0.0110)	(0.0110)	
In Interviewer Experience			× ,	-0.0325***	0.0003
-				(0.0081)	(0.0331)
Panel C: Time-Constant Effects of In	ndividual-Level Controls			· · · ·	· · · ·
Male	-0.1657***	-0.1667***	-0.1667***	-0.1667***	-0.1563***
	(0.0098)	(0.0098)	(0.0098)	(0.0098)	(0.0094)
German-Born	-0.0468***	-0.0343**	-0.0340**	-0.0347**	-0.0353**
	(0.0151)	(0.0152)	(0.0152)	(0.0152)	(0.0152)
Observations	382,012	382,012	382,012	382,012	382,012
Number of Individuals	49,756	49,756	49,756	49,756	49,756
R^2 Within	0.0891	0.0899	0.0899	0.0901	0.103
R^2 Overall	0.292	0.293	0.293	0.293	0.329
R^2 Between	0.407	0.407	0.407	0.407	0.462
Individual Fixed Effects	Yes	Yes	Yes	Yes	Yes
State-Year Fixed Effects	Yes	Yes	Yes	Yes	Yes
Educational Status	Yes	Yes	Yes	Yes	Yes
Marital Status	Yes	Yes	Yes	Yes	Yes
Number of Children	Yes	Yes	Yes	Yes	Yes
Birth Decade Fixed Effects	Yes	Yes	Yes	Yes	Yes
Panel Attrition Control	Yes	Yes	Yes	Yes	Yes
Survey-Type Fixed Effects	Yes	Yes	Yes	Yes	Yes
Interviewer ID Fixed Effects	No	No	No	No	Yes
Interviewer ID Fixed Effects	INU	INU	110	110	165

Table 9: Within and Between Effects of Age on SWB – Cubic Age Function

Notes: The dependent variable is self-reported individual well-being, measured from Low = 0 to High = 10. In Net Income is the log of the monthly household net income (in EUR). Employed refers to an indicator variable that takes the value 1 for individuals who are currently employed. Health Status is the current self-rated health status, measured as very good=1 to bad=5. In Survey Experience is the log of the number of years a particular individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for individuals appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 for individual appears in the survey. Interviewer Gender is an indicator variable that takes the value 1 if the interviewer is male. In Interviewer Experience is the log of the number of years a particular interviewer conducts interviews in the SOEP. Male takes the value 1 for males. German-Born takes the value 1 for individuals born in Germany or German immigrants prior to the year 1949. State-Year Fixed Effects refer to a set of Federal-State-Year Time Fixed Effects. Education Status refer to a set of indicator variables for individuals with educational status less than high school or completed high school education. Marital Status refer to a set of indicator variables for individuals being single, married, separated, or divorced. No. of Children refers to the number of children in the household. Birth Decade Fixed Effects refer to a set of indicator variables for each birth decade in the survey. Constant term included but not shown. Panel Attrition Control refers to an indicator variable which is 1 if the individual does not participate in the next wave of the survey. Survey-Type Fixed Effects refer to a set of indicator variables with respect to the type of survey (e.g., phone, self-filled (with/without interviewer), verbal and/or written, with/without accompanying person but with interpreter. ID Fixed Effects refer to a set of indicator variables according to the u

*: Significant at the 10% level. **: Significant at the 5% level. ***: Significant at the 1% level.

B Descriptive Statistics

Table 10: Summary Statistics for the Main Regression	n Variables of the German SOEP Household Survey
--	---

		-			•
Variable	Ν	Mean	SD	Minimum	Maximum
SWB	382012	7.0906	1.7521	0	10
Age	382012	4.8218	1.6918	1.6000	10.5000
Age squared	382012	26.1117	17.3310	2.5600	110.2500
Age cubed	382012	154.7460	147.7413	4.0960	1157.6250
ln Net Income	382012	7.7376	0.568	2.3026	12.2061
Employed	382012	0.6137	0.4869	0	1
Health Status	382012	2.6182	0.9617	1	5
In Survey Experience	382012	1.9325	0.9326	0	3.4965
Interviewer Gender	382012	0.5683	0.4953	0	1
In Interviewer Experience	382012	11.3838	7.5512	1	33
Male	382012	0.4728	0.4993	0	1
German-Born	382012	0.8482	0.3588	0	1
Number of Children	382012	0.6656	1.0228	0	11
Marital: Single	382012	0.2044	0.4033	0	1
Marital: Married	382012	0.6347	0.4815	0	1
Marital: Separated	382012	0.0226	0.1487	0	1
Marital: Divorced	382012	0.0741	0.2619	0	1
Marital: Widowed	382012	0.0642	0.2451	0	1
Education: Below High School	382012	0.1930	0.3947	0	1
Education: High School	382012	0.6124	0.4872	0	1
Education: Above High School	382012	0.1946	0.3959	0	1

										One summer of	NUMBER OF THE PARTY OF THE PART	for the memory which matrice and to extend the memory and the meter of an and the second									
			Age	Age		German	-		Marital Status:	tus:		Number of	Health	ln Net		Education	Educational Status relative to High School	to High School	ln Survey	Interviewer Interviewer	Interviewer
Variable	SWB	Age	squared	cubed	Male	Born	Single	Married	Separated	Divorced	Widowed	children	Status	Income	Employed	Below	High School	Above	Experience	Gender	Experience
SWB	1.0000																				
Age	-0.0617	1.0000																			
Age squared	-0.0565	0.9837	1.0000																		
Age cubed	-0.0513	0.9443	0.9877	1.0000																	
Male	-0.0014	-0.0051	-0.0080	-0.0119	1.0000																
German-Born	-0.0100	0.0912	0.1015	0.1058	-0.0045	5 1.0000															
Marital: Single	0.0039	-0.5008	-0.4264	-0.3600	0.0530	0.0827	1.0000														
Marital: Married	0.0758	0.2095	0.1420	0.0846	0.0613	-0.1064	1 -0.6681	1.0000													
Marital: Separated	-0.0513	-0.0029	-0.0111	-0.0164	-0.0167	7 -0.0218	-0.0771	-0.2006	1.0000												
Marital: Divorced	-0.0698	0.0438	0.0196	-0.0005	-0.0608	8 0.0293	-0.1434	-0.3729	-0.0431	1.0000											
Marital: Widowed	-0.0497	0.3673	0.4084	0.4367	-0.1326	5 0.0549	-0.1327	-0.3452	-0.0399	-0.0741	1.0000										
Number of children	0.0677	-0.3745	-0.3960	-0.3916	-0.0176	3 -0.1445	6.1240	0.1982	0.0105	-0.0424	-0.1465	1.0000									
Health Status	-0.4267	0.3819	0.3750	0.3595	-0.0472	2 0.0141	-0.1900	0.0582	0.0097	0.0378	0.1520	-0.1545	1.0000								
ln Net Income	0.2489	-0.1066	-0.1399	-0.1628	0.0709	0.0671	-0.0952	0.2873	-0.0716	-0.1494	-0.2048	0.1859	-0.1785	1.0000							
Employed	0.0799	-0.4142	-0.4754	-0.5064	0.1293	0.0150	0.0730	0.0310	0.0226	0.0450	-0.2427	0.1664	-0.2379	0.2725	1.0000						
Education: Below High School	-0.0673	-0.0209	0.0157	0.0409	-0.1068	8 -0.2417	0.0504	-0.0823	0.0038	-0.0174	0.0951	0.0221	0.0762	-0.2072	-0.1958	1.0000					
Education: High School	-0.0300	-0.0133	-0.0241	-0.0304	0.0369	0.1663	-0.0125	0.0134	-0.0053	0.0212	-0.0252	-0.0313	0.0104	-0.0750	0.0604	-0.6147	1.0000				
Education: Above High School	0.1040	0.0372	0.0140	-0.0034	0.0611	0.0362	-0.0349	0.0656	0.0027	-0.0087	-0.0638	0.0165	-0.0888	0.2989	0.1208	-0.2404	-0.6178	1.0000			
ln Survey Experience	-0.1154	0.1666	0.1688	0.1640	0.0156	0.0765	-0.0306	0.0042	0.0056	-0.0178	0.0576	-0.1803	0.1152	-0.0164	-0.0305	0.0373	-0.0178	-0.0153	1.0000		
Interviewer Gender	0.0028	0.0145	0.0142	0.0135	0.0047	0.0216	0.0023	-0.0038	-0.0104	0.0009	0.001	-0.001	-0.0067	-0.0041	-0.0110	-0.0032	0.0092	-0.0082	-0.0384	1.0000	
Interviewer Experience	-0.0276	0.0914	0.0939	0.0930	0.0051	0.0093	-0.0058	-0.0131	0.0099	0.0039	0.0251	-0.1139	0.0388	0.0722	-0.0185	0.0053	-0.0370	0.0403	0.4436	-0.0432	1.0000

C Figures



Figure 1: Relationship Between Subjective Well-Being and Age – Linear Age Function

Notes: This figure shows the *within*-effect and *between*-effect of age on subjective well-being based on the coefficient estimates in Table 7. See the main text for additional details on the estimation method and the set of included individual level controls.



Figure 2: Relationship Between Subjective Well-Being and Age – Squared Age Function

Notes: This figure shows the *within*-effect and *between*-effect of age on subjective well-being based on the coefficient estimates in Table 8. See the main text for additional details on the estimation method and the set of included individual level controls.



Figure 3: Relationship Between Subjective Well-Being and Age – Cubed Age Function

Notes: This figure shows the *within*-effect and *between*-effect of age on subjective well-being based on the coefficient estimates in Table 9. See the main text for additional details on the estimation method and the set of included individual level controls.

D Data Description

=

Table 12: Data Description f	or the Main Regression	Variables of the German S	OEP Household Survey

Variable	Description		
SWB	Life satisfaction measured on an 11 point scale. Answer to the question: "How satisfied an		
	you at present with your life, all things considered? Please respond using the following scale		
	where '0' indicates not at all satisfied and '10' indicates completely satisfied"		
Age	Age of individual in $(1 \text{ unit} = 10 \text{ years})$. Generated variable in SOEP, by substraction of		
	birth year from survey year.		
Net Income	Total monthly net income of household.		
Employed	Dummy variable indicating whether the respondent is employed $(=1)$ or undemployed $(=0)$		
	at the time of the interview.		
Health Status	Self-reported individual health status rated on a five point Likert scale from 1 (very good		
	to 5 (bad).		
Survey Experience	Total times the individual has taken part in the survey.		
Interviewer Gender	Gender of person who interviewed the respondent.		
Interviewer Experience	Total times the interviewer was part of the survey team and actually interviewed a responden		
	Interviewer experience is independent from survey experience since interviewers may chang		
	across time.		
Male	Dummy variable indicating whether the respondent is male $(=1)$ or female $(=0)$.		
German-Born	Dummy variable taking a value of 1 for individuals born in Germany or if the individual		
	immigrated to Germany prior to the year 1949, and zero otherwise.		
No of Children	The total number of children living in the household at the time of the interview.		
Marital: Single	Dummy variable indicating whether a person is single $(=1)$ at the time of the interview an		
	has not been married before.		
Marital: Married	Dummy variable indicating whether a person is married $(=1)$ at the time of the interview.		
Marital: Separated	Dummy variable indicating whether a person is married and separated from his/her partne		
	(=1) at the time of the interview.		
Marital: Divorced	Dummy variable indicating whether a person is divorced $(=1)$ at the time of the interview		
Marital: Widowed	Dummy variable indicating whether a person is married and the partner has died $(=1)$.		
Education: Below High School	Dummy variable indicating whether the highest degree of education a person has achieved		
	below high school level.		
Education: High School	Dummy variable indicating whether the highest degree of education a person has achieved i		
	equal to high school level.		
Education: Above High School	Dummy variable indicating whether the highest degree of education a person has achieved		
	higher than high school level.		

Zuletzt erschienen /previous publications:

V-421-19	Philipp Biermann, Jürgen Bitzer, Erkan Gören : The Relationship between Age and Subjective Well-Being: Estimating Within and Between Effects Simultaneously
V-420-19	Philipp Poppitz, Multidimensional Inequality and Divergence: The Eurozone Crisis
V-419-19	in Retrospect Heinz Welsch, Utilitarian and Ideological Determinants of Attitudes toward
	Immigration: Germany before and after the "Refugee Crisis"
V-418-19	Christoph Böhringer, Xaquin Garcia-Muros, Mikel González-Eguino, Greener
	and Fairer: A Progressive Environmental Tax Reform for Spain
V-417-19	Heinz Welsch, Martin Binder, Ann-Kathrin Blankenberg, Pro-environmental norms and subjective well-being: panel evidence from the UK
V-416-18	Jasper N. Meya, Environmental Inequality and Economic Valuation
V-415-18	Christoph Böhringer, Thomas F. Rutherford, Edward J. Balistreri, Quantifying
	Disruptive Trade Policies
V-414-18	Oliver Richters, Andreas Siemoneit, The contested concept of growth imperatives:
	Technology and the fear of stagnation
V-413-18	Carsten Helm, Mathias Mier, Subsidising Renewables but Taxing Storage?
	Second-Best Policies with Imperfect Carbon Pricing
V-412-18	Mathias Mier, Policy Implications of a World with Renewables, Limited
	Dispatchability, and Fixed Load
V-411-18	Klaus Eisenack, Mathias Mier, Peak-load Pricing with Different Types of
	Dispatchability
V-410-18	Christoph Böhringer, Nicholas Rivers, The energy efficiency rebound effect in
	general equilibrium
V-409-18	Oliver Richters, Erhard Glötzl, Modeling economic forces, power relations, and
	stock-flow consistency: a general constrained dynamics approach
V-408-18	Bernhard C. Dannemann, Erkan Gören, The Educational Burden of ADHD:
	Evidence From Student Achievement Test Scores
V-407-18	Jürgen Bitzer, Erkan Gören, Foreign Aid and Subnational Development: A Grid Cell Analysis
V-406-17	Christoph Böhringer, Jan Schneider, Marco Springmann, Economic and
	Environmental Impacts of Raising Revenues for Climate Finance from Public Sources
V-405-17	Erhard Glötzl, Florentin Glötzl, Oliver Richters, From constrained optimization to
	constrained dynamics: extending analogies between economics and mechanics
V-404-17	Heinz, Welsch, Jan Kühling, How Green Self Image Affects Subjective Well-Being:
	Pro-Environmental Values as a Social Norm
V-403-17	Achim Hagen, Jan Schneider, Boon or Bane? Trade Sanctions and the Stability of
	International Environmental Agreements
V-402-17	Erkan Gören, The Role of Novelty-Seeking Traits in Contemporary Knowledge
	Creation
V-401-17	Heinz Welsch, Jan Kühling, Divided We Stand: Immigration Attitudes, Identity, and
	Subjective Well-Being
V-400-17	Christoph Böhringer, Thomas F. Rutherford, Paris after Trump: An inconvenient
	insight
V-399-17	Frank Pothen, Heinz Welsch, Economic Development and Material Use
V-398-17	Klaus Eisenack, Marius Paschen, Designing long-lived investments under uncertain
	and ongoing change
V-397-16	Marius Paschen, The effect of intermittent renewable supply on the forward
	premium in German electricity markets
V-396-16	Heinz Welsch, Philipp Biermann, Poverty is a Public Bad: Panel Evidence from
	Subjective Well-being Data
V-395-16	Philipp Biermann, How Fuel Poverty Affects Subjective Well-Being: Panel
	Evidence from Germany
V-394-16	Heinz Welsch, Electricity Externalities, Siting, and the Energy Mix: A Survey
V-393-16	Leonhard Kähler, Klaus Eisenack, Strategic Complements in International

Environmental Agreements: a New Island of Stability

V-392-16	Christoph Böhringer, Xaquin Garcia-Muros, Ignacio Cazcarro, Iñaki Arto , The Efficiency Cost of Protective Measures in Climate Policy
V-391-16	Achim Hagen, Juan-Carlos Altamirano-Cabrera, Hans-Peter Weikard, The Influence of Political Pressure Groups on the Stability of International Environmental Agreements
V-390-16	Christoph Böhringer, Florian Landis, Miguel Angel Tovar Reaños, Cost- effectiveness and Incidence of Renewable Energy Promotion in Germany
V-389-16	Carsten Helm, Mathias Mier, Efficient diffusion of renewable energies: A roller- coaster ride
V-388-16	Christoph Böhringer, Jan Schneider, Emmanuel Asane-Otoo, Trade In Carbon and The Effectiveness of Carbon Tariffs
V-387-16	Achim Hagen, Leonhard Kähler, Klaus Eisenack, Transnational Environmental Agreements with Heterogeneous Actors
V-386-15	Jürgen Bitzer, Erkan Gören, Sanne Hiller, Absorption of Foreign Knowledge: Firms' Benefits of Employing Immigrants
V-385-15	Klaus Eisenack, Julien Minnemann, Paul Neetzow, Felix Reutter, Contributions to the institutional economics of the energy transition
V-384-15	Christoph Böhringer, Xaquín Garcia-Muros, Mikel Gonzalez-Eguino, Luis Rey, US Climate Policy: A Critical Assessment of Intensity Standards
V-383-15	Christoph Böhringer, Edward J. Balistreri, Thomas F. Rutherford, Carbon policy and the structure of global trade
V-382-15	Christoph Böhringer, Brita Bye, Taran Fæhn, Knut Einar Rosendahl, Output- based rebating of carbon taxes in the neighbor's backyard
V-381-15	Christoph Böhringer, Markus Bortolamedi, Sense and No(n)-Sense of Energy Security Indicators
V-380-15	Christoph Böhringer, Knut Einar Rosendahl, Halvor Briseid Storrøsten, Mitigating carbon leakage: Combining output-based rebating with a consumption tax
V-379-15	Jan Micha Steinhäuser, Klaus Eisenack, Spatial incidence of large-scale power
V-378-15	plant curtailment costs Carsten Helm, Franz Wirl , Climate policies with private information: The case for unilateral action
V-377-15	Klaus Eisenack , Institutional adaptation to cooling water scarcity in the electricity sector under global warming
V-376-15	Christoph Böhringer, Brita Bye, Taran Fæhn, and Knut Einar Rosendahl,
V-375-15	Targeted carbon tariffs – Carbon leakage and welfare effects Heinz Welsch, Philipp Biermann, Measuring Nuclear Power Plant Externalities Using Life Satisfaction Data: A Spatial Analysis for Switzerland
V-374-15	Erkan Gören , The Relationship Between Novelty-Seeking Traits And Comparative Economic Development